% EJC papers *must* begin with the following two lines. 
\documentclass[12pt]{article}
\usepackage{e-jc}

% Please remove all other commands that change parameters such as
% margins or pagesizes.

% only use standard LaTeX packages
% only include packages that you actually need

% we recommend these ams packages
\usepackage{amsthm,amsmath,amssymb}
% we recommend the graphicx package for importing figures
\usepackage{graphicx}
\usepackage{enumerate}
\usepackage{multicol} 

% use this command to create hyperlinks (optional and recommended)
\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}

% use these commands for typesetting doi and arXiv references in the bibliography
\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}

% all overfull boxes must be fixed; 
% i.e. there must be no text protruding into the margins


% declare theorem-like environments
\theoremstyle{plain}
\newtheorem{theorem}{Theorem}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{fact}[theorem]{Fact}
\newtheorem{observation}[theorem]{Observation}
\newtheorem{claim}[theorem]{Claim}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{open}[theorem]{Open Problem}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{question}[theorem]{Question}

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{note}[theorem]{Note}

\renewcommand{\labelenumi}{{\rm(\roman{enumi})}}


\newcommand{\de}{\em}
\newcommand{\ds}{\displaystyle}

\newcommand{\Real}{\mathbb{R}}
\newcommand{\Complex}{\mathbb{C}}
\newcommand{\Rational}{\mathbb{Q}}
\newcommand{\Natural}{\mathbb{N}}
\newcommand{\Integer}{\mathbb{Z}}
\newcommand{\Sphere}{S}

\newcommand{\QED}{\hfill\qed}
\newcommand{\KER}[1]{\mathrm{Ker}\,{#1}}
\newcommand{\IM}[1]{\mathrm{Im}\,{#1}}
\newcommand{\abs}[1]{\left\vert{#1}\right\vert}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

% if needed include a line break (\\) at an appropriate place in the title

\title{\bf On spherical designs of some harmonic indices}

% input author, affilliation, address and support information as follows;
% the address should include the country, and does not have to include
% the street address 


\author{  \hspace{0.1cm} Yan Zhu \hspace{0.8cm} Eiichi Bannai\\
\small School of Mathematical Sciences  \\[-0.8ex]
\small  Shanghai Jiao Tong University\\[-0.8ex] 
\small Shanghai, 200240, China\\
\small\tt \{zhuyan,bannai\}@sjtu.edu.cn\\
\and
\hspace{-0.6cm}Etsuko Bannai \hspace{1.6cm} Kyoung-Tark Kim  \hspace{1.4cm} Wei-Hsuan Yu\\
\small  \hspace{0.4cm}Misakigaoka 2-8-21 \hspace{0.7cm} School of Mathematical Sciences  \hspace{0.3cm} Department of Mathematics  \\[-0.8ex]
\small  \hspace{0.7cm}Itoshima-shi \hspace{1.5cm}  Shanghai Jiao Tong University \hspace{0.7cm}  Michigan State University\\[-0.8ex]
\small \hspace{-1.2cm}Fukuoka, 819-1136, Japan \hspace{0.7cm} Shanghai, 200240, China  \hspace{1.8cm} MI, 48824, U.S.A.\\
\small\tt  \hspace{-1.3cm} et-ban@rc4.so-net.ne.jp\hspace{0.6cm}  kyoungtarkkim@gmail.com \hspace{1.1cm} u690604@gmail.com\\}

% \author{ \hspace{0.5cm} Yan Zhu \hspace{4.5cm} Eiichi Bannai\\
% \small School of Mathematical Sciences \hspace{1.5cm} School of Mathematical Sciences \\[-0.8ex]
% \small  Shanghai Jiao Tong University \hspace{1.8cm}  Shanghai Jiao Tong University\\[-0.8ex] 
% \small Shanghai, 200240, China \hspace{4.4cm} Shanghai, China\\
% \small\tt zhuyan@sjtu.edu.cn \hspace{3.2cm}  bannai@sjtu.edu.cn\\
% \and
%  \hspace{1cm}Etsuko Bannai \hspace{4cm} Kyoung-Tark Kim\\
% \small \hspace{0.9cm} Misakigaoka 2-8-21 \hspace{3.4cm} School of Mathematical Sciences \\[-0.8ex]
% \small \hspace{1.3cm} Itoshima-shi \hspace{4.2cm}  Shanghai Jiao Tong University\\[-0.8ex] 
% \small \hspace{-1cm}Fukuoka, 819-1136, Japan \hspace{4.2cm} Shanghai, China\\
% \small\tt et-ban@rc4.so-net.ne.jp\hspace{2.8cm}  kyoungtarkkim@gmail.com\\
% \and
% Wei-Hsuan Yu\\
%  \small Department of Mathematics  \\[-0.8ex]
%  \small  Michigan State University\\[-0.8ex] 
%  \small MI, U.S.A.\\
%  \small\tt u690604@gmail.com\\}

% \date{\dateline{submission date}{acceptance date}\\
% \small Mathematics Subject Classifications: comma separated list of
% MSC codes available from http://www.ams.org/mathscinet/freeTools.html}

\date{\dateline{September 3, 2016}{XX}\\
\small Mathematics Subject Classifications: 05B30, 11D09, 12E10}

\begin{document}
\begin{thebibliography}{10}
\bibitem{BV2008}
 C. Bachoc, F. Vallentin.
\newblock {\em  New upper bounds for kissing numbers from semidefinite programming}.
\newblock J. Amer. Math. Soc., 21 (2008), no. 3, 909--924.

\bibitem{BOT2015}
E. Bannai, T. Okuda and M. Tagami.
\newblock {\em  Spherical designs of harmonic index $t$}.
\newblock J. Approx. Theory, 195 (2015), 1--18.

\bibitem{BY2013}
A. Barg and W.-H. Yu.
\newblock {\em  New bounds for spherical two-distance sets}.
\newblock Exp. Math., 22 (2013), no. 2, 187--194.

\bibitem{BDHSS2010}
P.~G. Boyvalenkov, P.~D. Dragnev, D.~P. Hardin, E.~B. Saff and M.~M. Stoyanova.
\newblock {\em   Universal lower bounds for potential energy of spherical codes}.
\newblock Constr. Approx., (2016). DOI: 10.1007/s00365-016-9327-5.


\bibitem{D1973}
P. Delsarte.
\newblock {\em  An algebraic approach to the association schemes of coding theory}.
\newblock Thesis, Universite Catholique de Louvain (1973),
Philips Res. Repts Suppl., 10 (1973).

\bibitem{DGS1977}
 P. Delsarte, J.~M. Goethals and J.~J. Seidel.
\newblock {\em Spherical codes and designs}.
\newblock  Geometriae Dedicata, 6 (1977), no. 3, 363--388.

\bibitem{DS1989}
 P. Delsarte and J.~J. Seidel.
\newblock {\em Fisher type inequalities for Euclidean $t$-designs}.
\newblock Linear Algebra Appl., 114/115 (1989), 213--230.


\bibitem{DN2010}
D.~K. Dimitrov and G.~P. Nikolov.
\newblock {\em Sharp bounds for the extreme zeros of classical orthogonal polynomials}.
\newblock  J. Approx. Theory. 162 (2010), no. 10, 1793--1804.

\bibitem{EL1992}
A. Elbert and A. Laforgia.
\newblock {\em Asymptotic formulas for ultraspherical polynomials $P^{\lambda}_n(x)$ and their zeros for large values of $\lambda$}.
\newblock  Proc. Amer. Math. Soc., 114 (1992), no. 2, 371--377.

\bibitem{G2005}
D. Gijswijt.
\newblock {\em  Matrix Algebras and Semidefinite Programming Techniques for Codes}. 
\newblock arXiv:1007.0906v1.

\bibitem{GL2003}
C. Giordano and A. Laforgia.
\newblock {\em On the Bernstein-type inequalities for ultraspherical polynomials}.
\newblock  Proceedings of the 6th International Symposium on Orthogonal Polynomials, Special Functions and their Applications (Rome, 2001). J. Comput. Appl. Math.,  153 (2003), no. 1--2, 243--248.

\bibitem{LRS1977}
D.~G. Larman, C.~A. Rogers and J.~J. Seidel.
\newblock {\em On two-distance sets in Euclidean space}.
\newblock Bull. London Math. Soc., 9 (1977), no. 3,  261--267.

\bibitem{LS1973}
P.~W.~H. Lemmens and J.~J. Seidel.
\newblock {\em  Equiangular lines}.
\newblock J. Algebra, 24 (1973), 494--512.

\bibitem{LV1993}
Y.~I. Lyubich and L.~N. Vaserstein.
\newblock {\em Isometric embeddings between classical Banach spaces, cubature formulas, and spherical designs}.
\newblock Geom. Dedicata, 47 (1993), no. 3, 327--362.

\bibitem{M2009}
 O.~R. Musin.
\newblock {\em Spherical two-distance sets}.
\newblock J. Combin. Theory Ser. A., 116 (2009), no. 4, 988--995.

\bibitem{N2011}
H. Nozaki.
\newblock {\em  A generalization of Larman-Rogers-Seidel's theorem}.
\newblock Discrete Math., 311 (2011), no. 10-11, 792--799. 

\bibitem{OY2015}
 T. Okuda and W.-H. Yu.
\newblock {\em A new relative bound for equiangular lines and nonexistence of tight spherical designs of harmonic index 4}.
\newblock European J. Combin., 53 (2016), 96--103.


\bibitem{SAGE}
SageMath, \verb@http://www.sagemath.org@.

\bibitem{S2003} 
G. Szeg\H{o}.  
\newblock {\em Orthogonal Polynomials}.  
\newblock 4th edition, Amer. Math. Soc., Colloquim Publications, volume 23 (2003).

\end{thebibliography}


\maketitle


% E-JC papers must include an abstract. The abstract should consist of a
% succinct statement of background followed by a listing of the
% principal new results that are to be found in the paper. The abstract
% should be informative, clear, and as complete as possible. Phrases
% like "we investigate..." or "we study..." should be kept to a minimum
% in favor of "we prove that..."  or "we show that...".  Do not
% include equation numbers, unexpanded citations (such as "[23]"), or
% any other references to things in the paper that are not defined in
% the abstract. The abstract will be distributed without the rest of the
% paper so it must be entirely self-contained.

\begin{abstract}
A finite subset $Y$ on the unit sphere $\Sphere^{n-1} \subseteq \mathbb{R}^n$ is called a spherical design of harmonic index $t$, if the following condition is satisfied:
$\sum_{\mathbf{x}\in Y}f(\mathbf{x})=0$ for all real homogeneous harmonic polynomials $f(x_1,\ldots,x_n)$ of degree $t$.
Also, for a subset $T$ of $\mathbb{N} = \{1,2,\cdots \}$, a finite subset $Y \subseteq \Sphere^{n-1}$ is called a spherical design of harmonic index $T,$ if $\sum_{\mathbf{x}\in Y}f(\mathbf{x})=0$ is satisfied for all real homogeneous harmonic polynomials $f(x_1,\ldots,x_n)$ of degree $k$ with $k\in T$.

In the present paper we first study Fisher type lower bounds for the sizes of spherical designs of harmonic index $t$ (or for harmonic index $T$).
We also study `tight' spherical designs of harmonic index $t$ or index $T$.
Here `tight' means that the size of $Y$ attains the lower bound for this Fisher type inequality.
The classification problem of tight spherical designs of harmonic index $t$ was started by Bannai-Okuda-Tagami (2015), and the case $t = 4$ was completed by Okuda-Yu (2016).
In this paper we show the classification (non-existence) of tight spherical designs of harmonic index 6 and 8, as well as the asymptotic non-existence of tight spherical designs of harmonic index $2e$ for general $e\geq 3$.
We also study the existence problem for tight spherical designs of harmonic index $T$ for some $T$, in particular, including index $T = \{8,4\}$.

  % keywords are optional


  \bigskip\noindent \textbf{Keywords:} spherical designs of harmonic index;
 Gegenbauer polynomial; Fisher type lower bound; tight design;
 Larman-Rogers-Seidel's theorem; Delsarte's method; semidefinite programming; elliptic diophantine equation.
\end{abstract}

\section{Introduction and spherical designs of harmonic index $t$ (or $T$)}
\label{sec:introduction}

\noindent
Throughout this paper $Y$ is assumed to be a finite non-empty set, and we denote the set of positive (resp. non-negative) integers by $\mathbb{N}$ (resp. $\mathbb{N}_0$).

Let ${\Sphere^{n-1} = \{\mathbf{x} = (x_1,\cdots ,x_n)\in \mathbb{R}^n \, \mid\, x_1^2 + \cdots +x_n^2=1\}}$ be the unit sphere in the Euclidean space $\mathbb{R}^n.$
Delsarte-Goethals-Seidel \cite[Definition 5.1]{DGS1977} (1977) gave the following definition of spherical designs.

\begin{definition}[Spherical $t$-designs]\label{dfe:1.1}
Let $ {t\in \mathbb{N}_0}$.
A subset $Y\subseteq \Sphere^{n-1}$ is called a \textit{spherical $t$-design} on $\Sphere^{n-1}$, if
\begin{equation}\label{spht}
\frac{1}{\left|\Sphere^{n-1}\right|}\int_{\mathbf{x}\in\Sphere^{n-1}}f(\mathbf{x})d\sigma(\mathbf{x})=\frac{1}{\left|Y\right|}\sum_{\mathbf{y}\in Y}f(\mathbf{y})
\end{equation}
for any real polynomial $f(x_1,\ldots ,x_n)$ of degree at most $t$, where $\left|\Sphere^{n-1}\right|$ denotes the volume (or the surface area) of the sphere $\Sphere^{n-1},$
and the integral is the surface integral on $\Sphere^{n-1}.$
\end{definition}

The condition~\eqref{spht} is known to be equivalent to the condition:
\[ \sum_{\mathbf{y}\in Y}f(\mathbf{y})=0 \; \mbox{ for all }  f(x_1, \ldots, x_n)\in \mathrm{Harm}_k^n,\; 1\leq k\leq t\, , \]
where $\mathrm{Harm}_k^n$ is the space of real homogeneous harmonic polynomials of degree $k$ in $n$ indeterminates.

In connection with the latter equivalent defining condition for spherical $t$-designs, we define a weaker concept which we call designs of harmonic index $t$ as follows.

\begin{definition}[Spherical designs of harmonic index $t$]\label{dfe:1.2}
A subset $Y\subseteq  \Sphere^{n-1}$ is called a \textit{spherical design of harmonic index $t$} on $\Sphere^{n-1}$, if
\begin{equation}\label{equ:star} \tag{$\ast$}
\sum_{\mathbf{y}\in Y}f(\mathbf{y})=0
\end{equation}
for all real homogeneous harmonic polynomial $f(x_1,\ldots ,x_n)$ of degree exactly $t$.
\end{definition}

More generally we have the following definition.

\begin{definition}[Spherical designs of harmonic index $T$]\label{dfe:1.3}
Let $T$ be a subset of $\mathbb{N}$.
A subset $Y\subseteq  \Sphere^{n-1}$ is called a \textit{spherical design of harmonic index $T$} on $\Sphere^{n-1}$ if
\[\sum_{\mathbf{y}\in Y}f(\mathbf{y})=0 \; \mbox{ for all } f(x_1,\ldots,x_n) \in \mathrm{Harm}_k^n \mbox{ with } k\in T.\]
\end{definition}

A spherical design of harmonic index $T$ with $T=\{1,2,\ldots ,t\}$ corresponds to a usual spherical $t$-design, and the case $T=\{t\}$ corresponds to a spherical design of harmonic index $t$.

We should remark that the concept of spherical designs of harmonic index $t$ (or $T$) was already introduced by Delsarte-Seidel \cite[Definition 4.1]{DS1989} (1989).
However the study of this topic is started in Bannai-Okuda-Tagami \cite{BOT2015} (2015).

The purpose of this paper is to study spherical designs of harmonic index $t$ as well as harmonic index $T$ for some $T$, and to convince the reader that these are interesting mathematical objects.
Our main concerns are Fisher type lower bounds for spherical designs of harmonic index $t$ and $T$, as well as the classification problems of so-called `tight' designs.
Here `tight' means those that attain the lower bound in a Fisher type inequality.
In Section~\ref{sec:LP} we provide a linear programming bound for spherical designs of harmonic index $T$.
We also formulate Fisher type inequalities and tight spherical designs of harmonic index $t$ or $T$.
In Section~\ref{sec:tf}, we discuss our philosophy how to find our test functions.
In the subsequent sections we will study some specific problems.
In Section~\ref{sec:nonexistence} the complete non-existence results for tight spherical designs of harmonic index 6 and 8 are proved.
Note that the case of $t=4$ was already settled by Okuda-Yu \cite{OY2015} in a beautiful way by applying the SDP (semidefinite programming) to the existence problem of equiangular lines.
Also note that our proofs for $t=6$ and $t=8$ are obtained in an elementary level without recourse to such deeper consideration as SDP.
In Section~\ref{sec:asymptotic} we show the asymptotic non-existence of harmonic index $2e$ case for general $e\geq 3$.
Then we turn our attention to the case of $T=\{t_1, t_2\}$.
The central model is the case $T=\{8,4\}$ in Section~\ref{sec:84}.
In Section~\ref{sec:several} we study the cases $T= \{8,2\}$, $\{8,6\}$, $\{6,2\}$, $\{6,4\}$, as well as $\{10,6,2\}$, and $\{12,8,4\}$.
We conclude the paper in Section~\ref{sec:remark} with some remarks.

The techniques which we used in the present paper are:
(i) the linear programming method by Delsarte, (ii) the detailed information on the locations of the zeros as well as the local minimum values of Gegenbauer polynomials, (iii) the generalization by Nozaki of the Larman-Rogers-Seidel theorem on $2$-distance sets to $s$-distance sets, (iv) the theory of elliptic diophantine equations, and (v) the semidefinite programming method of eliminating some $2$-angular line systems for small dimensions.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Linear programming method for spherical designs of harmonic index $T$}
\label{sec:LP}

\noindent
In this section we consider a linear programming bound for spherical designs of harmonic index $T$.
We also introduce some terminology and notation which will be used in the subsequent sections.

Let $Q_{n,k}(x)$ be the \textit{Gegenbauer polynomial} of degree $k$ in one variable $x$ as introduced in \cite[Definition 2.1]{DGS1977}.
Recall how the polynomials $Q_{n,k}(x)$ are normalized \cite[Theorem 2.4, Theorem 3.2]{DGS1977}:
\[ Q_{n,k}(1) = \mathrm{dim}\, \mathrm{Harm}_k^n = \binom{n+k-1}{n-1} - \binom{n+k-3}{n-1} =:h_{n,k} \, . \]
The Gegenbauer polynomials $Q_{n,k}(x)$ are orthogonal polynomials on the closed interval $[-1,1]$ with respect to the weight function $(1-x^2)^{(n-3)/2}$, i.e.,
\[ \int_{-1}^{1} Q_{n,k}(x) \, Q_{n,\ell}(x) \, (1-x^2)^{\frac{n-3}{2}}\; dx = a_{n,k} \delta_{k,\ell} \, ,\]
where $a_{n,k}$ is some (normalization) constant depending on $n$ and $k$, and $\delta_{k,\ell}$ is the Kronecker delta.
From this orthogonality it is well established that to any real polynomial $F(x)$ of degree $r$ we can associate its \textit{Gegenbauer expansion}
\begin{equation}\label{eq:Gexp}
 F(x) = \sum_{k=0}^{r} f_k \, Q_{n,k}(x)\, ,
\end{equation}
where the \textit{Gegenbauer coefficients $f_k$} can be computed as follows:
\[ f_k = \frac{1}{a_{n,k}} \int_{-1}^{1} F(x)\, Q_{n,k}(x)\, (1-x^2)^{\frac{n-3}{2}} \; dx.\]

We denote by $\mathbf{x}\cdot \mathbf{y}$ the standard inner product of $\mathbf{x}$ and $\mathbf{y}$ in $\mathbb{R}^n$.
For a subset $Y \subseteq \mathbb{R}^n$ we set $I(Y) := \{ \mathbf{x}\cdot \mathbf{y} \,\mid \, \mathbf{x},\mathbf{y} \in Y,\; \mathbf{x}\neq \mathbf{y}\}$.
If $\{ e_{k,1}, \ldots, e_{k,h_{n,k}} \}$ is any orthonormal basis for $\mathrm{Harm}_k^n$ with respect to the inner product $\langle f,g\rangle = \frac{1}{\left|\Sphere^{n-1}\right|} \int_{\mathbf{x}\in\Sphere^{n-1}} f(\mathbf{x})g(\mathbf{x}) d\sigma(\mathbf{x})$ in $\mathrm{Harm}_k^n$, then the well-known \textit{addition formula} says that, for every $\mathbf{x},\mathbf{y} \in \Sphere^{n-1}$,
\[ Q_{n,k}(\mathbf{x}\cdot \mathbf{y}) = \sum_{i=1}^{h_{n,k}} e_{k,i}(\mathbf{x}) e_{k,i}(\mathbf{y}).\]
From the addition formula we have, for any $Y\subseteq \Sphere^{n-1}$,
\[ \begin{array}{rcl}
M_k(Y) & := & {\displaystyle \sum_{\mathbf{x},\mathbf{y} \in Y} Q_{n,k}(\mathbf{x}\cdot \mathbf{y}) = \sum_{\mathbf{x},\mathbf{y} \in Y} \sum_{i=1}^{h_{n,k}} e_{k,i}(\mathbf{x}) e_{k,i}(\mathbf{y}) } \\
& & \\
 & = & {\displaystyle \sum_{i=1}^{h_{n,k}} \sum_{\mathbf{x},\mathbf{y} \in Y} e_{k,i}(\mathbf{x}) e_{k,i}(\mathbf{y}) = \sum_{i=1}^{h_{n,k}} \left( \sum_{\mathbf{x} \in Y} e_{k,i}(\mathbf{x}) \right)^2 }. \\
\end{array} \]
Thus we obtain (see Definition~\ref{dfe:1.3} for (M2)) the following two simple observations:
\begin{enumerate}
\item [(M1)] The quantity $M_k(Y)$ is always non-negative;
\item [(M2)] Moreover, $M_k(Y) = 0$ for any $k\in T$ if and only if $Y \subseteq \Sphere^{n-1}$ is a spherical design of harmonic index $T$.
\end{enumerate}

We introduce our main identity (see \eqref{eq:mi} below).
(See \cite{DGS1977} for the original discussion about so-called `\textit{linear programming bounds}' for spherical designs.)
Suppose $F(x)$ is a non-constant real polynomial of degree $r$ which is of the form \eqref{eq:Gexp}.
For any $Y \subseteq \Sphere^{n-1}$, if we calculate $\sum_{\mathbf{x},\mathbf{y} \in Y} F(\mathbf{x}\cdot \mathbf{y})$ in two different ways, we find
\begin{equation}\label{eq:mi}
\left| Y \right| F(1) + \sum_{\mathbf{x},\mathbf{y}\in Y, \atop \mathbf{x}\neq \mathbf{y}} F(\mathbf{x}\cdot \mathbf{y}) = \left| Y \right|^2 f_0 + \sum_{k=1}^r f_k \, M_k(Y).
\end{equation}

Now suppose $Y \subseteq \Sphere^{n-1}$ is a spherical design of harmonic index $T$.
We are interested in finding a good lower bound for $\left| Y \right|$.
(If $Y_1, Y_2 \subseteq \Sphere^{n-1}$ are spherical designs of harmonic index $T$ with $Y_1 \cap Y_2 = \emptyset$, then so is $Y_1 \cup Y_2$.)
If $F(x)$ satisfies
\begin{enumerate}
\item [(LP1)] $F(u) \geq 0$ for each $u \in [-1,1]$;
\item [(LP2)] $f_k \leq 0$ for each non-negative integer $k$ not in $T$.
\end{enumerate}
then we obtain (recall (M1) and (M2)) that
\begin{equation}\label{eq:lpe}
\left| Y \right| F(1) \leq \mbox{LHS of \eqref{eq:mi}}=\mbox{RHS of \eqref{eq:mi}} \leq \left| Y \right|^2 f_0 \, ,
\end{equation}
where the first and second inequalities are due to (LP1) and (LP2), respectively.
Moreover we necessarily have $f_0 >0$, since by (LP1) the integrand of the following integral
\[ f_0 = \frac{1}{a_{n,0}} \int_{-1}^{1} F(x) \, (1-x^2)^{\frac{n-3}{2}}\; dx \]
is non-negative and $a_{n,0} = \int_{-1}^{1} (1-x^2)^{\frac{n-3}{2}}\, dx > 0$.
By \eqref{eq:lpe} we have
\begin{equation}\label{eq:mie}
\left| Y \right| \geq \frac{F(1)}{f_0}.
\end{equation}
Therefore we want to find the following quantity:
\begin{equation}\label{eq:LP}
S := \sup \left\{ \frac{F(1)}{f_0} \mid F(x) \mbox{ satisfies (LP1) and (LP2)} \right\}.
\end{equation}
\begin{definition}
We call a polynomial $F(x)$ a \textit{test function} provided that both (LP1) and (LP2) hold.
\end{definition}
Note that the multiplication of $F(x)$ by any \emph{positive} real number does not affect the consistency of (LP1) and (LP2), and does not change the value $\frac{F(1)}{f_0}$.
Thus we might adopt an \textit{equivalence relation} on the set of test functions defined using the multiplication by a positive real number.

It is also harmless to restrict the range of the above supremum by only considering $F(x)$ with (any) fixed constant term $f_0 = c$.
If we introduce new variables $g_k := \frac{f_k}{f_0}$ ($k\geq 0$), then the computation of $S$ in \eqref{eq:LP} becomes the following \textit{linear programming} of infinite variables $(g_1, g_2, \ldots)$.
(Notice that \eqref{eq:LP} itself is not a linear programming of the variables $(f_0, f_1, \ldots)$ because $f_0$ appears in the denominator of $\frac{F(1)}{f_0}$ so that $\frac{F(1)}{f_0}$ is not actually linear in $f_0$.)
\begin{equation}\label{eq:LPR}
S = \sup \left\{ 1 + \sum_{k=1}^{\infty} g_k \, Q_{n,k}(1) \right\} \quad \mbox{ subject to }
\end{equation}
\begin{enumerate}
\item All but finite number of $g_1, g_2, \ldots \in \Real$ are zero;
\item A polynomial $G(x) = 1 + \sum_{k=1}^{\infty} g_k \, Q_{n,k}(x)$ satisfies (LP1);
\item All $g_1, g_2, \ldots$ satisfy (LP2).
\end{enumerate}


\section{How do we find a good test function?}\label{sec:tf}
\noindent In this paper we consider the case where $T$ consists of positive even integers $t_1>t_2>\cdots >t_\ell\geq 2$.
Given $n$ and $T$, what is the \emph{best} choice of a test function $F(x)$ for \eqref{eq:LP}?
The answer is not easy:
First of all, what is the definition of ``\textit{goodness}'' for $F(x)$?
Of course, the most obvious one is that a ``better'' $F(x)$ provides a larger $\frac{F(1)}{f_0}$.
However, this approach seems hopeless because the exact determination of $S$ in \eqref{eq:LP} is too difficult.
(It is not even apparent that there exists an optimal test function which attains the supremum in \eqref{eq:LP}.)
This means that there can be many approaches for choosing a test function with different definitions of ``goodness''.

In this section we show our way for choosing a test function, and explain its philosophy.
We should emphasize here that our goal is not to maximize $\frac{F(1)}{f_0}$.
In this paper, for our purpose, we only deal with a test function $F(x)$ of the following form:
\begin{equation}\label{eq:tfunction}
F(x) = f_0 + \sum_{k \in T} f_{k} \, Q_{n,k}(x).
\end{equation}
One reason why we only concern test functions of the form \eqref{eq:tfunction} is that it is the easiest one we can handle.
More precisely, Eq.~\eqref{eq:tfunction} is the easiest form for a polynomial $F(x)$ to satisfy (LP2).
Another reason is that if $F(x)$ has the form \eqref{eq:tfunction}, then we are able to guess when the equality in \eqref{eq:mie} is attained:

\begin{proposition}\label{prop:obs}
Let $Y\subseteq \Sphere^{n-1}$ be a spherical design of harmonic index $T$.
Suppose a polynomial $F(x)$ has the form \eqref{eq:tfunction}, and satisfies $\mathrm{(LP1)}$.
$($Thus $F(x)$ becomes a test function.$)$
Then, $\left\vert Y \right\vert = \frac{F(1)}{f_0}$ if and only if $F(\alpha) = 0$ for all $\alpha \in I(Y)$.
\end{proposition}

\begin{proof}
Recall $\left\vert Y \right\vert \geq \frac{F(1)}{f_0}$.
Thus, $\left\vert Y \right\vert = \frac{F(1)}{f_0}$ if and only if equality in \eqref{eq:lpe} holds if and only if ${\displaystyle \sum_{\mathbf{x},\mathbf{y}\in Y, \atop \mathbf{x}\neq \mathbf{y}} F(\mathbf{x}\cdot \mathbf{y}) = 0 = \sum_{k\in T} f_k \, M_k(Y)}$.
Suppose $\left\vert Y \right\vert = \frac{F(1)}{f_0}$.
Since $F(x)$ has the form \eqref{eq:tfunction} and $M_k(Y) = 0$ for all $k \in T$ (Definition~\ref{dfe:1.3}), the term $\sum\limits_{k \in T} f_k M_k(Y)$ vanishes.
We see from (LP1) that the equation ${\displaystyle \sum_{\mathbf{x},\mathbf{y}\in Y, \atop \mathbf{x}\neq \mathbf{y}} F(\mathbf{x}\cdot \mathbf{y}) = 0}$ is equivalent to that $F(\alpha) = 0$ for all $\alpha \in I(Y)$.
\end{proof}

Thus, if $Y \subseteq \Sphere^{n-1}$ is a spherical design of harmonic index $T = \{ t_1 > \cdots > t_{\ell} \}$ ($t_i$ : all even) and $F(x)$ is a test function of the form \eqref{eq:tfunction}, then the previous proposition says that $\left\vert Y \right\vert = \frac{F(1)}{f_0}$ if and only if $Y$ is a \emph{distance set} where the distances between two distinct points in $Y$ should occur in the zero set of $F(x)$.

Now it is the time when we should explain our definition of ``goodness'' for $F(x)$.
Our philosophy for choosing $F(x)$ (of the form \eqref{eq:tfunction}) is that we want a distance set $Y$ with large $|I(Y)|$ whenever $\left\vert Y \right\vert$ attains $\frac{F(1)}{f_0}$, i.e., we hope an \emph{interesting $($=large$)$} ``tight'' object.
Therefore we require that
\begin{equation}\label{eq:cond}
\mbox{ a test function } F(x) \mbox{ has exactly } \ell \mbox{ non-negative zeros}
\end{equation}
because we think that $\ell$ is the naturally expected largest number of non-negative zeros for $F(x)$ of the form \eqref{eq:tfunction}.
(Of course, it is still a difficult problem to determine the precise maximum number of minima for $F(x)$ in \eqref{eq:tfunction}.)

We only deal with the case where there are only finitely many test functions satisfying our conditions \eqref{eq:cond}, although the choices of test functions are possibly infinite for some index sets.
Suppose that there are only $m$ test functions $F_1(x),F_2(x),\ldots,F_m(x)$ (up to equivalence) which satisfy our condition \eqref{eq:cond} for a given index set $T$.
For each test function $F_i(x)$, put $b_{n,T,F_i} := \frac{F_i(1)}{f_0^{(i)}}$, where $f_0^{(i)}$ is the constant term in the Gegenbauer expansion of $F_i(x)$.
\begin{definition}\label{def:tight}
With the above conditions and notation, we define $b_{n,T} := \max\limits_{1\leq i\leq m}\{b_{n,T,F_i}\}$.
If the size of a spherical design $Y$ of harmonic index $T$ attains the lower bound $b_{n,T}$, then $Y$ is said to be \textit{tight}.
\end{definition}
 If there exists a unique test function $F(x)$ (up to equivalence), then Definition \ref{def:tight} implies that $b_{n,T}=\frac{F(1)}{f_0}$.

\begin{remark}\label{rem:tight}
 We should note that it is a very delicate problem to define ``tight'' designs for general $T$ in a rigorous way.
In the above argument, we required that our test function should have as many zeros as possible so that $b_{n,T}$ could be as large as possible.
The reason to get as many zeros is that the ``tight'' set can be an $s$-distance set for larger $s$, otherwise
the size cannot be large.
\end{remark}

In the first section we defined the concept of spherical design of harmonic index $t$ or more general $T$.
This notion was already essentially defined in the literature, as ``a spherical design which admits indices $T$''.
(See Delsarte-Seidel \cite{DS1989}, say.)
On the other hand, the terminology of spherical design of harmonic index $t$ is already defined as ``a spherical design for which equality in Definition \ref{dfe:1.2} \eqref{equ:star} holds for any \emph{homogeneous polynomials of degree t}'', say in \cite{DS1989,LV1993}, etc.
In order to avoid the confusion with these terminologies, we use the term `spherical designs of harmonic index $t$ (or $T$)'.
It seems that no systematic study of spherical designs of harmonic index $T$ has been made, before Bannai-Okuda-Tagami \cite{BOT2015}.
They used the test function $F(x)=c_{n,t}+Q_{n,t}(x)$, with $c_{n,t}=-\min\limits_{-1\leq x\leq 1}Q_{n,t}(x)$ for $T=\{t\}$, and obtained the following theorem.
(Note that this theorem is a special case of Proposition \ref{prop:obs}.)
\begin{theorem}[{\cite[Theorem 1.2]{BOT2015}}]\label{theo:3.1}
Let $Y$ be a design of harmonic index $t$ on $S^{n-1}$.
Then the following inequality holds:
\begin{equation}\label{eq:3.1}
\left| Y \right| \geq 1+ \frac{Q_{n,t}(1)}{c_{n,t}},
\end{equation}
where $c_{n,t}=-\min\limits_{-1\leq x\leq 1}Q_{n,t}(x)$.
Moreover equality in \eqref{eq:3.1} holds if and only if
$Q_{n,t}(\alpha)=-c_{n,t}$ holds for any $\alpha\in I(Y)$.
\end{theorem}
\begin{remark}\label{rem:unique}
For $T=\{t\}$, the choice of test function is unique (up to equivalence).
The reason is that our test function is of form \eqref{eq:tfunction}, i.e. $F(x)=f_0+Q_{n,t}(x)$.
 And $f_0=-\min\limits_{-1\leq x\leq 1}Q_{n,t}(x)$ is uniquely determined, from the monotonicity of the local minima of the Gegenbauer polynomial.
Hence $b_{n,\{t\}}=1+\frac{Q_{n,t}(1)}{c_{n,t}}$ and we write $b_{n,t}=b_{n,\{t\}}$ for simplicity.
\end{remark}
In \cite{BOT2015, OY2015} they discussed the cases when $t=2$ and $4$.
In the following section we discuss the existence problem of tight spherical designs of harmonic index $t$ for $t=6$ and $8$.
As is seen from Remark \ref{rem:unique}, if we consider $T=\{2e\}$, then there is only one positive zero of $F(x)$, equivalently, only one positive minima of $Q_{n,2e}(x)$ exists.

If we take $T=\{8,4\}$, say, there are only at most two positive zeros.
With property \eqref{eq:cond} and some calculation, it is shown that our test function $F(x)$ is determined uniquely.
For $T=\{t_1,t_2,\ldots,t_\ell\}$ with $t_i$ even, it seems that we can expect that there are at most $\ell$ non-negative zeros although we do not know the exact answer for general $T$.
Among the possible test functions with this property, finding the best one, namely with the largest $b_{n,T}$, is not easy for general $T$.
For example, in the discussion below, in the case of $T=\{12,8,4\}$ the candidates of test function $F(x)$ are not necessarily unique.
Also, in some cases, no good test function exists.
Thus, here we are compromising in taking the test function which seems to be the most natural one.
We cannot eliminate the possibility of the existence of a better test function, in general case.
For specific $T$, which are discussed in subsequent sections, we believe the choices of our test functions are natural and meaningful, although we do not show that rigorously at this stage.
This situation may look to be an embarrassing situation, but this is even true for the definition of tight spherical $t$-designs, originally defined by Delsarte-Goethals-Seidel \cite{DGS1977}.
In the case of ordinary tight spherical $2e$-designs, the specifically chosen test function
$(R_e(x))^2=(Q_{n,0}(x)+\cdots +Q_{n,e}(x))^2$
satisfies the requirement that there are possible maximum $e$ positive zeros.
Moreover, it is expected to give the maximum $b_{n,T}$ (with $T=\{1,2,\ldots,2e\}$), so it satisfies our criterion of `good' test function.
On the other hand, there is no easy proof that it is the best test function.
Still, the concept of tight spherical $t$-design in this particular choice of the test function was very meaningful.
Our definition of tight spherical designs of harmonic index $T$ has the same feature, and we have to compromise that the definitions of tight designs are not completely rigorously defined in the general case of $T$.
(To show the test function is a best one is not easy and in many cases it is still undecided.
However, such a problem is an unavoidable fact in this kind of theories.)


\section{The non-existence of tight spherical designs of
harmonic index 6 and 8}
\label{sec:nonexistence}

\noindent
In this section we will prove the non-existence of tight spherical designs of harmonic index $t=6$ and $t=8$.
The lower bound in Theorem \ref{theo:3.1} is
obtained by the inequality~\eqref{eq:mie} using following test function $F(x)$.
\begin{equation}\label{eq:cnt}
F(x) = c_{n,t} + Q_{n,t}(x) \quad\mbox{with} \quad c_{n,t} := -\min Q_{n,t}(x).
\end{equation}
Throughout this section, $F(x)=c_{n,2e}+Q_{n,2e}(x)$ and
we say
$Y$ is a tight spherical design of harmonic index $2e$ if $|Y|$ attains the lower bound $b_{n,2e}$ in \eqref{eq:3.1}.
\begin{lemma}\label{lem:upbd}
If $Y$ is a tight spherical design of harmonic index $2e$, then $|Y| \leq \frac{n(n+1)}{2}$.
\end{lemma}
\begin{proof}
Recall that if $Y\subseteq \Sphere^{n-1}$ is a tight spherical design of harmonic index $t$, then $I(Y) \subseteq \{x \,\mid\, F(x)=0\}$.
It is known that $F(x)$ have exactly two roots $\{\alpha,-\alpha\}$, which means that $Y$ is bounded above by the cardinality of spherical 2-distance set.
For any spherical $2$-distance set $X \subseteq \Sphere^{n-1}$ with $I(X) = \{ \alpha, \beta \}$ and $\alpha+\beta \geq 0$, Musin \cite{M2009} proved $\left| X \right| \leq \frac{n(n+1)}{2}$.
Therefore $|Y|\leq \frac{n(n+1)}{2}$.
\end{proof}

\subsection{The non-existence of tight spherical designs of harmonic index $6$}
$ $\\
In this subsection $Y$ denotes a tight spherical design of harmonic index $6$.
The Gegenbauer polynomial $Q_{n,6}(x)$
(with our normalization $Q_{n,6}(1)=\dim \mbox{Harm}_6^n$) is given by
\[\begin{array}{rcl}
Q_{n,6}(x) & = & {\scriptstyle \frac{n(n+2)(n+10)}{6!} \left\{ (n+4)(n+6)(n+8)x^6 -15(n+4)(n+6)x^4 + 45(n+4)x^2-15 \right\} }.\\
\end{array} \]
By taking the largest root for $Q_{n,6}^{\prime}(x)=0$, we get (see e.g. the proof of \cite[Corollary 4.1]{BOT2015}) the point $\alpha$ at which $Q_{n,6}(x)$ takes the minimum value, i.e., $Q_{n,6}(\alpha) = -c_{n,6}$.
The lower bound $b_{n,6}$ of $|Y|$ defined in \eqref{eq:3.1}
can be obtained as well.
The following are our results:
\[ \begin{array}{l}
\alpha^2=\frac{5(n+6)+\sqrt{10(n+3)(n+6)}}{(n+6)(n+8)}, \\
c_{n,6}=-\frac{n(n+2)(n+10) \left( 2(n-2)(n+3)(n+6)+(n+3)(n+4)\sqrt{10(n+3)(n+6)}\right)}{36(n+6)(n+8)^2}, \\
b_{n,6}=\frac{(n+4)\left( 20\sqrt{10(n+3)(n+6)}+(n+3)(n+6)(n^2+9n-12) \right)}{20\left( 2(n-2)(n+6)+(n+4)\sqrt{10(n+3)(n+6)} \right)}.
\end{array}\]

It is not difficult to check that $|Y|=b_{n,6}> \frac{n(n+1)}{2}$ if $n\geq 37$.
Moreover, $b_{2,6}=2$ and $b_{24,6}=231$ are the only two cases for which $b_{n,6} \in \mathbb{Z}$ when $n \leq 36$.

\begin{remark}\label{remark:4.2}
We should remark that not all the roots of $F(x)$ in \eqref{eq:cnt} will necessarily appear in $I(Y)$ when $n$ is small.
Consider the case $n=2$.
Recall that $b_{2,2e} = 2$ is proved for general $e$ in \cite[p.~6]{BOT2015}.
Let $y_1,y_2$ be two unit vectors in $\Real^2$ with angle $\theta =j \pi/{2e}$ for odd $j$.
Then, by the argument in \cite[p.~2]{BOT2015}, $Y=\{ y_1, y_2 \}$ is a tight spherical design of harmonic index $2e$ on $\Sphere^1$.
\end{remark}

Larman-Rogers-
Seidel (1977) proved the following fact.
\begin{theorem}[{\cite[Theorem 2]{LRS1977}}]\label{theo:4.5}
Let $X$ be a $2$-distance set in $\mathbb{R}^n$ with Euclidean distances $c$ and $d$ $(c < d)$.
If $|X|>2n+3$, then we have
\[ \frac{c^2}{d^2} = \frac{(k-1)}{k} \]
for some integer $k$ with $2 \leq k \leq \frac{1+\sqrt{2n}}{2}$.
\end{theorem}

Suppose $n=24$.
Then $Y$ is an at most $2$-distance set in $\Sphere^{23}$ with $I(Y) \subseteq \{ \pm \alpha\}$.
Assume $X$ is a spherical 2-distance set with $I(X)=\{\pm \alpha\}$ such that $|Y|\leq |X|$. (There exists such $X$, otherwise $|Y|$ is strictly larger than the cardinality of any 2-distance set.)
If we put $c=\sqrt{2-2\alpha}$ and $d=\sqrt{2+2\alpha}$, then $c$ and $d$ become the Euclidean distances between two distinct vectors in $X$.
Note that $|X|\geq |Y|=b_{24,6} = 231 > 2 \times 24+3 = 51$.
However, in this case, we obtain $c^2/d^2=1/3$ from easy calculation, contrary to Theorem~\ref{theo:4.5}.
Hence there exists no tight spherical design of harmonic index $6$ when $n=24$.



\subsection{The non-existence of tight spherical designs of harmonic index $8$}
$ $\\
In this subsection $Y \subseteq \Sphere^{n-1}$ denotes a tight spherical design of harmonic index $8$.
The Gegenbauer polynomial $Q_{n,8}(x)$ is
\[ \begin{array}{rcl}
Q_{n,8}(x)&=& {\scriptstyle \frac{n(n+2)(n+4)(n+14)}{8!} \big\{ (n+6)(n+8)(n+10)(n+12)x^8 -28(n+6)(n+8)(n+10)x^6} \\
& & \quad\qquad\qquad\qquad {\scriptstyle +210(n+6)(n+8)x^4 - 420(n+6)x^2+105\big\} }.\\
\end{array} \]
As in the preceding subsection we can obtain $\alpha$, $c_{n,8}$, and also
\begin{equation}\label{eq:asymbn8}
b_{n,8}=\frac{1}{252 \times 12.03144913\cdots}\times n^4(1+o(1)).
\end{equation}
It can be checked that $\left| Y \right| = b_{n,8} > \frac{n(n+1)}{2}$ if $n \geq 20$ and, if $n \leq 19$,  the only integral value is $b_{2,8}=2$.
By a similar argument as in Remark~\ref{remark:4.2} one trivial example exists when $n=2$.
\begin{remark}
We do not give the formulas of $\alpha$, $b_{n,8}$ and $c_{n,8}$ explicitly, since they are extremely complicated.
Here, $b_{n,8} > \frac{n(n+1)}{2}$ is checked from the formula of $b_{n,8}$ rather than from the asymptotic form \eqref{eq:asymbn8}.
\end{remark}


\section{The asymptotic non-existence of tight spherical designs of harmonic index $2e$ for general $e$}
\label{sec:asymptotic}

\noindent
In this section we consider the existence of tight spherical designs of harmonic index $2e$ for $e\geq 5$, since the cases $e=2,3,4$ were already treated.
Our main result in this section is the following theorem.

\begin{theorem}\label{theo:5.1}
Let $e \geq 2$ be fixed.
Then there exist positive constants $A_{2e}$ and $B_{2e}$ such that
\[\lim_{n\rightarrow \infty}\frac{c_{n, 2e}}{n^e}=A_{2e} \quad\mbox{ and }\quad \ \lim_{n\rightarrow \infty}\frac{b_{n, 2e}}{n^e}=B_{2e},\]
where $A_{2e}$ and $B_{2e}$ depend only on $e$.
Therefore,
\[c_{n,2e}=A_{2e}n^e(1+o(1)) \quad\mbox{ and }\quad \ b_{n,2e}=B_{2e}n^e(1+o(1)).\]
\end{theorem}

\begin{corollary}
Let $e \geq 3$ be fixed.
If $n$ is sufficiently large, then there exist no tight spherical designs of harmonic index $2e$.
\end{corollary}

\begin{proof}
If $Y$ is a tight spherical design of harmonic index $2e$, then $I(Y) \subseteq \{\pm \alpha\}$ for some $\alpha>0$, and it follows from Lemma \ref{lem:upbd} that $\left| Y \right| \leq \frac{n(n+1)}{2}$.
On the other hand, if $n$ is sufficiently large, then Theorem~\ref{theo:5.1} implies
\[ \left| Y \right| = b_{n,2e}=B_{2e}n^e(1+o(1)),\]
a contradiction.
\end{proof}

\begin{proof}[Proof of Theorem~\ref{theo:5.1}]
Szeg\H{o} \cite[p.~107]{S2003} gives the asymptotic property of Gengebauer polynomial $C_t^{\lambda}(x)$:
\[\lim_{\lambda\rightarrow \infty} \lambda^{-\frac{t}{2}}C_t^{\lambda}(\lambda^{-\frac{1}{2}} x) = \frac{H_{t}(x)}{t!},\]
where $H_t(x)$ is the Hermite polynomial of degree $t$.

Recall that if $n\geq 3$ then $Q_{n,t}(x)=\frac{n+2t-2}{n-2}C_t^{(n-2)/2}(x)$.
(See e.g. \cite[p.~365]{DGS1977}.)
Putting $\lambda=\frac{n-2}{2}$ and $t = 2e$, we have
\[ \begin{array}{rcl}
{\displaystyle \lim_{\lambda\rightarrow \infty} \lambda^{-\frac{t}{2}}C_t^{\lambda}(\lambda^{-\frac{1}{2}}x)} & = & {\displaystyle \lim_{n\rightarrow \infty} \left( \frac{n-2}{2}\right)^{-e}\frac{n-2}{n+4e-2} \, Q_{n,2e}(\sqrt{\frac{2}{n-2}}\, x ) } \\
 & & \\
 & = & {\displaystyle 2^e\lim_{n\rightarrow \infty} n^{-e}Q_{n,2e}(\sqrt{\frac{2}{n-2}}\, x)}.
\end{array} \]
Set $P_{n,e}(x)=n^{-e}Q_{n,2e}(\sqrt{\frac{2}{n-2}}\, x)$ for simplicity.
Then we have
\begin{equation}\label{equ:5.1}
2^e\lim_{n\rightarrow \infty} P_{n,e}(x)=\frac{H_{2e}(x)}{(2e)!}.
\end{equation}
Take the derivative with respect to $x$ on both sides of (\ref{equ:5.1}).
For fixed $e$, since $P^{\prime}_{n,e}(x)$ uniformly converges to $\frac{2^e}{(2e-1)!}x^{2e-1}$ as $n$ tends to be infinity, we get the following result:
\[
2^e \lim_{n\rightarrow \infty} \frac{d}{dx} P_{n,e}(x)=2^e\frac{d}{dx} \left( {\lim_{n\rightarrow \infty} P_{n,e}(x)} \right)=\frac{H^{\prime}_{2e}(x)}{(2e)!}=\frac{4e}{(2e)!}{H_{2e-1}(x)},
\]
where the last equality is due to the property $H_t^{\prime}(x)=2tH_{t-1}(x)$.
Let $x_1$ be the largest zero of $H_{2e-1}(x)$.
Then
\[2^e \lim_{n\rightarrow \infty}  \frac{d}{dx}\left( n^{-e}Q_{n,2e}(\sqrt{\frac{2}{n-2}}\, x)\right) \Big|_{x=x_1}=\frac{1}{(2e)!}H^{\prime}_{2e}(x_1)=0.\]
Thus the following equality can be obtained.
\[A_{2e}=-\lim_{n\rightarrow \infty}\frac{\min Q_{n, 2e}(x)}{n^e}=-\lim_{n\rightarrow \infty}\frac{ Q_{n, 2e}(\sqrt{\frac{2}{n-2}}\, x_1)}{n^e}=-\frac{H_{2e}(x_1)}{2^e (2e)!}.\]
Recall that $b_{n,t}=1+\frac{Q_{n,t}(1)}{c_{n,t}}$ and $Q_{n,t}(1) = \binom{n+t-1}{n-1} - \binom{n+t-3}{n-1}$. This implies
\[B_{2e}=\lim_{n\rightarrow \infty}\frac{b_{n, 2e}}{n^e}=\frac{1}{(2e)!A_{2e}}=-\frac{2^e}{H_{2e}(x_1)}.\]
\end{proof}

\begin{remark}
In Theorem~\ref{theo:5.1} we did not give explicit evaluation of $B_{2e}$, but it is possible to give it, since the locations of the zeros of Hermite polynomials and the (local) minimum values of $H_{2e}(x)$ are well studied.
Also, if we want to evaluate $b_{n,2e}$ explicitly from below, rather than evaluating $B_{2e}$, it is also possible, although we will not discuss it in this paper.
For this purpose, the following papers \cite{BDHSS2010}, \cite{DN2010}, \cite{EL1992}, \cite{GL2003}  may be useful to do that.
It seems that there are many literature on this.
\end{remark}

\section{Tight spherical designs of harmonic index $\{8,4\}$}
\label{sec:84}

\noindent
In what follows, we assume $T=\{t_1,t_2,\ldots,t_\ell\}$ with $t_1=2e>t_2> \cdots >t_\ell$ and $t_i$ ($1\leq i \leq \ell$) even.
And we investigate the case where our test function is of the form
$$F(x)=f_0+Q_{n,2e}(x)+f_{t_2}Q_{n,t_2}(x)+\cdots+f_{t_\ell}Q_{n,t_\ell}(x).$$
Let $L(x)=Q_{n,2e}(x)+f_{t_2}Q_{n,t_2}(x)+\cdots+f_{t_\ell}Q_{n,t_\ell}(x)$.
Then as we mentioned in Section \ref{sec:tf}, it is better that $F(x)$ have as many zeros in $[-1,1]$.
For this purpose
we are interested in the case where $L(x)$ attains minimum value at $\ell$ non-negative points $\alpha_1,\ldots,\alpha_\ell\in [0,1]$.

Delsarte-Goethals-Seidel (1977) gave an upper bound for a spherical $s$-distance set $X\subseteq \Sphere^{n-1}$.
\begin{theorem}[{\cite[Theorem 4.8]{DGS1977}}]\label{theo:s-dis set}
If $X$ is a spherical $s$-distance set in $\Sphere^{n-1}$, then $\left| X \right| \leq \binom{n+s-1}{n-1}+\binom{n+s-2}{n-1}$.
\end{theorem}
Using the above theorem, we obtain the following lemma which gives the relation between the size of tight spherical design of harmonic index $T$ and the cardinality of spherical $s$-distance set.
\begin{lemma}
If $Y$ is a tight spherical design of harmonic index $T$, and $L(x)=Q_{n,2e}(x)+f_{t_2}Q_{n,t_2}(x)+\cdots+f_{t_\ell}Q_{n,t_\ell}(x)$ takes minimum
value at $\ell$ non-negative points, then
$$\left| Y \right| \leq
\left\{\begin{array}{ll}\medskip
\binom{n+2\ell-2}{n-1}+\binom{n+2\ell-3}{n-1} &\mbox{if~} 2 \nmid e\\
\binom{n+2\ell-1}{n-1}+\binom{n+2\ell-2}{n-1}& \mbox{if~} 2\mid e
\end{array}\right.$$
\end{lemma}
\begin{proof}
Let $c_{n,T,L}=-\min L(x)$ and $F(x)=L(x)+c_{n,T,L}$.
If $e$ is even (resp. odd), then $F(x)$ has $2\ell$ (resp. $2\ell-1$) zeros.
By the assumption, $F(x)$ has $\ell$ non-negative roots.
Namely, $|I(Y)|\leq 2\ell$ if $e$ is even, then $Y\in \Sphere^{n-1}$ is an at most $2\ell$-distance set.
It follows from Theorem \ref{theo:s-dis set} that $\left| Y \right|\leq \binom{n+2\ell-1}{n-1}+\binom{n+2\ell-2}{n-1}$.
Similarly, we can get the conclusion if $e$ is odd.
\end{proof}


Consider the case of $T=\{t_1,\, t_2\}$ with $t_2=$ even and $t_1=2e > t_2$.
In this case we use the test function
$F(x)=f_0+Q_{n,2e}(x)+f_{t_2}Q_{n, t_2}(x)$.
Let $L(x)=Q_{n,2e}(x)+f_{t_2}Q_{n, t_2}(x)$.
In this case
we are interested in the cases when
$L(x)$ has the minimum value
$-c_{n,T}$ at exactly two non-negative points
$\alpha, \beta$ (with $\alpha>\beta$) and choose
$f_0=c_{n,T}$. For this purpose we take
\[L(x)=Q_{n,2e}(x)+f_{t_2}Q_{n, t_2}(x),\]
and we want to determine $f_{t_2}$ such that
\[ Q_{n,2e}(x)+f_{t_2}Q_{n,t_2}(x)=a (x^2-\alpha^2)^2 (x^2-\beta^2)^2-c_{n,T} \]
for some $\alpha, \beta$ and $c_{n,T}$.

Suppose $t_1=8$ and $t_2=4$.
Then the problem is to find $f_4$ such that
\begin{equation}\label{equ:6.1}
Q_{n,8}(x)+f_4Q_{n,4}(x)=a(x^2-\alpha^2)^2(x^2-\beta^2)^2-c_{n,T}.
\end{equation}
The Gegenbauer polynomial $Q_{n,4}(x)$ is
\[Q_{n,4}(x)=\frac{n(n+6)}{4!} \left\{ (n+2)(n+4)x^4-6(n+2)x^2+3 \right\}.\]
By comparing the coefficients in \eqref{equ:6.1}, we obtain the following equations:
\[
\begin{array}{l}
{\scriptsize a=\frac{n(n+2)(n+4)(n+6)(n+8)(n+10)(n+12)(n+14)}{8!}, }\\
\\
{\scriptsize 2a(\alpha^2+\beta^2)=\frac{28n(n+2)(n+4)(n+6)(n+8)(n+10)(n+14)}{8!}, }\\
 \\
{\scriptsize  a(\alpha^4+\beta^4+4\alpha^2\beta^2)=\frac{210n(n+2)(n+4)(n+6)(n+8)(n+14)}{8!} +\frac{n(n+2)(n+4)(n+6)}{4!}f_4,}\\
\\
{\scriptsize 2a\alpha^2\beta^2(\alpha^2+\beta^2)=\frac{420n(n+2)(n+4)(n+6)(n+14)}{8!}+\frac{6n(n+2)(n+6)}{4!}f_4, }\\
\\
{\scriptsize a\alpha^4\beta^4-c_{n,T}=\frac{105n(n+2)(n+4)(n+14)}{8!}+\frac{3n(n+6)}{4!}f_4. }
\end{array}
\]
Therefore,
\[ \begin{array}{l}
\ds \alpha^2+\beta^2=\frac{14}{n+12}, \\
 \\
\scriptsize (\alpha^2+\beta^2)^2+2\alpha^2\beta^2=\frac{210}{(n+10)(n+12)}+\frac{1680}{(n+8)(n+10)(n+12)(n+14)}f_4,\\
\\
\scriptsize \alpha^2\beta^2(\alpha^2+\beta^2)=\frac{210}{(n+8)(n+10)(n+12)}+\frac{5040}{(n+4)(n+8)(n+10)(n+12)(n+14)}f_4.
\end{array} \]
We have
\[ f_4=\frac{(n+4)(n+5)(n+14)}{60(n+12)}.\]
Hence $\alpha^2$ and $\beta^2$ are the solutions of the following quadratic equation in the variable $u$:
\[ u^2-\frac{14}{n+12}u +\frac{21}{(n+8)(n+12)}=0.\]
Finally, we obtain
\[
\begin{array}{l}
\ds \alpha^2,\beta^2=\frac{7(n+8) \pm 2\sqrt{7(n+5)(n+8)}}{(n+8)(n+12)},\\
\\
%\ds c_{n,T}=\frac{n(n+1)(n+4)(n+5)(n+10)(n+14)}{160(n+8)(n+12)},\\ \\
\ds b_{n,T}=\frac{1}{252}(n+1)(n+2)(n+5)(n+6),
\end{array}
\]
where $b_{n,T}$ is the lower bound in \eqref{eq:mie}.

In the following we shall prove the non-existence of tight spherical designs of harmonic index $\{8,4\}$ stated in Theorem \ref{theo:8-4design} below.
If $Y$ is a tight spherical design of harmonic index $\{8,4\}$, then $I(Y) \subseteq \{\pm \alpha, \pm \beta\}$.
We define
\[U(h) := \left\lfloor \; \frac{1}{2} + \sqrt{\frac{h^2}{2h-2}+\frac{1}{4}} \;\right\rfloor.\]
For a spherical $s$-distance set, Nozaki (2011) generalized Larman-Rogers-Seidel theorem \cite[Theorem 2]{LRS1977} as follows.
\begin{theorem}[{\cite[Theorem 5.1]{N2011}}]\label{theo:6.1}
Let $Y$ be an $s$-distance set on $\Sphere^{n-1}$ with $s \geq 2$ and $I(Y) = \{ \beta_1, \ldots, \beta_s \}$.
Put $N := \binom{n+s-2}{s-1} + \binom{n+s-3}{s-2}$.
If $\left| Y \right| \geq 2N$, then for each $i = 1,\dots,s$,
\[ k_i := \prod_{j=1,\dots,s, \atop j \neq i} \frac{1 - \beta_j}{\beta_i - \beta_j} \]
must be an integer with $\left| k_i \right| \leq U(N)$.
\end{theorem}

If $X=Y \cup (-Y)$ is an antipodal spherical $s$-distance set, then $Y$ is a spherical $(s-1)$-distance set.
Nozaki (2011) proved the following theorem.
The conditions of $\left| X \right|$ in Theorem~\ref{theo:6.2} are less restrictive than that in Theorem~\ref{theo:6.1}.

\begin{theorem}[{\cite[Theorem 5.2]{N2011}}]\label{theo:6.2}
Let $X$ be an antipodal $s$-distance set on $\Sphere^{n-1}$ where $s$ is an odd integer at least $5$. \\
Suppose $I(X)=\{-1,\pm\beta_1, \pm\beta_2, \ldots, \pm\beta_{\frac{s-1}{2}} \}$.
\begin{enumerate}[$(1)$]
\item Let $N=\binom{n+s-4}{s-3}$. If $\left| X \right| \geq 4N$, then for each $i = 1,\dots,(s-1)/2$,
\[ k_i := \prod_{j=1,\dots,\frac{s-1}{2}, \atop j \neq i} \frac{1 - \beta_j^2}{\beta_i^2 - \beta_j^2} \]
must be an integer with $\left| k_i \right| \leq U(N)$.
\item Let $N=\binom{n+s-3}{s-2}$. If $\left| X \right|\geq 4N+2$, then for each $i = 1,\dots,(s-1)/2$,
\[ k_i := \frac{1}{\beta_i} \prod_{j=1,\dots,\frac{s-1}{2}, \atop j \neq i} \frac{1 - \beta_j^2}{\beta_i^2 - \beta_j^2} \]
must be an integer with $\left| k_i \right| \leq \lfloor \sqrt{2N^2/(N+1)} \rfloor$.
\end{enumerate}
\end{theorem}

With the above theorem and {\cite[Theorem 5.3]{N2011}}, Nozaki showed that inner products for an antipodal spherical $s$-distance set are in fact rational.

\begin{theorem}[{\cite[Theorem 5.4]{N2011}}]\label{theo:6.3}
Suppose $X$ is an antipodal $s$-distance set on $\Sphere^{n-1}$ with $s \geq 4$.
If $|X| \geq 4\binom{n+s-3}{s-2}+2$, then $\beta$ is rational for
any $\beta \in I(X)$.
\end{theorem}

A tight spherical design of harmonic index $\{8,4\}$ is regarded as an at most 4-distance set $Y \subset \Sphere^{n-1}$ with $I(Y)\subseteq \{\pm\alpha, \pm\beta\}$.
We construct an antipodal set $X^{\prime}=Y \cup (-Y)$.
Note that $I(X^{\prime})\subseteq\{-1,\pm\alpha, \pm\beta\}$.
Assume $X\subset \Sphere^{n-1}$ is a spherical 5-distance set with $I(X)=\{-1,\pm\alpha, \pm\beta\}$ such that $|X|\geq |X^{\prime}|=2|Y|$.
By applying Theorem~\ref{theo:6.2} to the set $X$ for $s=5$, we obtain the next lemma.
\begin{lemma}\label{lem:rat}
Suppose $Y^{\prime}$ is a spherical $4$-distance set $\{\pm \alpha, \pm \beta\}$. Let $X=Y^{\prime} \cup (-Y^{\prime})$.
\begin{enumerate}[$(1)$]
\item If $ \left| Y^{\prime} \right| \geq 2\binom{n+1}{2}$, then the following two numbers are integers:
\[ k_1=\frac{1-\alpha^2}{\beta^2-\alpha^2}, \qquad k_2=\frac{1-\beta^2}{\alpha^2-\beta^2}. \]
\item If $\left| Y^{\prime}\right| \geq 2\binom{n+2}{3}+1$, then the following two numbers are integers:
\[ k_1=\frac{1-\alpha^2}{\beta(\beta^2-\alpha^2)}, \qquad k_2=\frac{1-\beta^2}{\alpha(\alpha^2-\beta^2)}. \]
\end{enumerate}
\end{lemma}
\begin{theorem}\label{theo:8-4design}
There exists no tight spherical design of harmonic index $\{8, 4\}$ on $S^{n-1}$ for all $n$.
\end{theorem}
\begin{proof}
If $Y$ is a tight spherical design of harmonic index $\{8,4\}$, then
\[ \left| Y \right|=\frac{(n+1)(n+2)(n+5)(n+6)}{252} \mbox{ with } I(Y)\subseteq \{ \pm \alpha, \pm \beta \},\]
where
\[\alpha,\beta=\sqrt{\frac{7(n+8) \pm 2\sqrt{7(n+5)(n+8)}}{(n+8)(n+12)}}.\]
Assume $X$ is a spherical 4-distance set with $I(X)=\{\pm\alpha,\pm\beta\}$ such that $|Y|\leq |X|$.
We shall consider three cases: $n\geq 76$, $9\leq n \leq 75$, and $2 \leq n \leq 8$.

\noindent \textbf{Case (1)}:
If $n \geq 76$, then
\[ {\textstyle |X|  \geq |Y|  \geq  2\left( \binom{n+2}{3}+\binom{n+1}{2} \right) > 2\binom{n+2}{3}+1 }. \]
By Theorem~\ref{theo:6.1}, $k_1$, $k_2$ are integers.
We have
\[k_2=\frac{1-\beta^2}{\alpha^2-\beta^2}= \frac{2+\sqrt{\frac{(n+5)(n+8)}{7}}}{4}=z \in \mathbb{Z}.\]
Hence $(n+5)(n+8)=7(4z-2)^2$.
By Lemma~\ref{lem:rat} we have
\[\alpha\beta = \sqrt{\frac {21} {(n+8)(n+12)}} \in \Rational.\]
Then $(n+8)(n+12)=21p^2/q^2$ for some coprime integers $p$ and $q$.
Furthermore $21p^2/q^2$ should be an integer.
Thus $q^2|21$, i.e., $q=1$.
We have $(n+8)(n+12)=21p^2$ and get the following table for some integers $y_1,y_2,y_3$.
\[ \begin{array}{c|c|c|c}
&n+5 & n+8 & n+12\\ \hline
\text{(i)} & y_1^2 & 7y_2^2 & 3y_3^2\\
\text{(ii)}& 7y_1^2 & y_2^2 & 21y_3^2\\
\text{(iii)} &3y_1^2 & 21y_2^2  & y_3^2\\
\text{(iv)} & 21y_1^2 & 3y_2^2 & 7y_3^2 \\ \hline
\end{array}\]

We know that $\gcd(n+5,n+8)=1$ or $3$.
If $\gcd(n+5,n+8)=1$, then $(n+5)(n+8)=7(4z-2)^2$ implies that $n+5=y_1^2$, $n+8=7y_2^2$ or $n+5=7y_1^2$, $n+8=y^2_2$.
If $\gcd(n+5,n+8)=3$, then $n+5=3y_1^2$, $n+8=21y_2^2$ or $n+5=21y_1^2$, $n+8=3y^2_2$.

For case (i), $n+12=3y_3^2$ is obtained from $(n+8)(n+12)=21p^2$.
We can similarly get the other three cases in the above table.\\
%Then we can prove the non-existence.
(i). $7=3y_3^2-y_1^2$ implies $y_1^2 \equiv 2 \; (\mathrm{mod} 3)$. Impossible.\\
(ii). $7=21y_3^2-7y_1^2$ implies $y_1^2 \equiv 2 \; (\mathrm{mod} 3)$. Impossible.\\
(iii). $3=21y_2^2-3y_1^2$ implies $y_1^2 \equiv 6 \; (\mathrm{mod} 7)$. Impossible.\\
(iv).  We cannot get a contradiction from a basic observation, but this problem can be formulated as the integral solutions of the following equation:
\[y^2=(n+5)(n+8)(n+12).\]
 By linear transformation $x=n+8$, this equation becomes $y^2=x^3+x^2-12x$.
From the database of elliptic curve with LMFDB label 168.b2, we know that $(x,y)=(-4,0),(0,0),(3,0)$ are all integral solutions of $y^2=x^3+x^2-12x$, namely, $y^2=(n+5)(n+8)(n+12)$ has no non-trivial integral solution.
\begin{remark}
$ $\\
i). Any elliptic curve over $\Rational$ has a Weierstrass equation of the form
\begin{equation}\label{equ:dstar} \tag{$\ast\ast$}
 y^2+a_1 x y+a_3 y = x^3+a_2 x^2 +a_4 x+a_6.
 \end{equation}
They are often displayed as a list $[a_1,a_2,a_3,a_4,a_6]$.
More information about the database of elliptic curve is available from:
\begin{center}
\verb@http://www.lmfdb.org/EllipticCurve/Q@
\end{center}

\noindent
ii).
The integral solutions of some elliptic equations of form \eqref{equ:dstar} can be solved using SAGE \cite{SAGE} with the following two commands (the reader should put suitable values for $a_1,a_2,a_3,a_4,a_6$):

\verb@E=EllipticCurve(QQ,[@$a_1,a_2,a_3,a_4,a_6$\verb@])@

\verb@E.integral_points()@
\end{remark}

\noindent \textbf{Case (2)}:
If $9 \leq n \leq 75$, then $|X| \geq |Y|  \geq  2\binom{n+1}{2}.$
Using the first statement in Lemma~\ref{lem:rat}, we see that both $\frac{1-\alpha^2}{\beta^2-\alpha^2}$ and $\frac{1-\beta^2}{\alpha^2-\beta^2}$ are integers.
It is easy to check that neither of them is an integer for $9 \leq n \leq 75$.

\noindent \textbf{Case (3)}:
If $2 \leq n \leq 8$, $b_{8,T}=65$ is the unique case with $b_{n,T} \in \Integer$. We set up the semidefinite programming (SDP) method on the upper bounds for spherical 4-distance sets with the indicated inner product values.
Theorem \ref{theo:SDP4d} is what we set up to estimate the upper bounds of spherical 4-distance sets.
 Such an SDP formula can be obtained from special setting of Bachoc-Vallentin \cite[p.~10--11]{BV2008} or generalization of Barg-Yu \cite[Theorem 3.1]{BY2013} for spherical $2$-distance sets.
We choose the positive semidefinite  matrices $S^n_k$  with size $(9-k) \times (9-k)$ and linear constraints $\sum_{c_i, c_j \in Y} Q_{n,k} ( \langle c_i, c_j \rangle) \geq 0 $ for $k=1, \cdots , 8$.
($S^n_k$ is the same notation in \cite{BY2013}).
When $n=8$, the SDP upper bound solved by CVX for the spherical 4-distance set is $50.23$.

We follow the argument in  \cite[p.~79]{G2005} to obtain rigorous bound of our semidefinite programming problems. We independently solve the dual problem to avoid numerical issue in CVX and guarantee that our bounds are justified. We checked that the error did not affect our results. i.e. our computational SDP bounds plus the error still strictly less than linear programming bound of tight spherical designs of harmonic index $T$.
Then, such tight designs do not exist.

In our problem, the solved value and the error are $1+\frac1 3\sum_{i=1}^4x_i=50.23$ and $\epsilon=\sum_{i=1}^{24} x_i \epsilon_i^\prime \leq 4.7605\times 10^{-5}$, where $\epsilon^\prime_i=\max\{\epsilon_i,0\}$.
 (Even if we take $\epsilon_i^ \prime=|\epsilon_i|$, the error $\epsilon=\sum_{i=1}^{24} x_i \epsilon_i^\prime$ is very small and bounded above by $0.00011592$.)
 Hence our computational SDP bound plus error is still strictly less than the LP bound (=65) for spherical design of harmonic index $\{8,4\}$.
We can conclude there exists no tight spherical design of harmonic index $\{8,4\}$.
\end{proof}

\begin{theorem}\label{theo:SDP4d}
Let $Y$ be a spherical $4$-distance set with inner products $a$, $b$, $c$ and $d$. Let $p$ be a positive integer. The
cardinality $|Y|$ is bounded above by the solution of the following semidefinite programming problem:

  \begin{equation*}
  1+\frac1 3 \max (x_1+ x_2+x_3+x_4)
  \end{equation*}
\centerline{\text{\rm subject to }}
  \begin{equation*}
\begin{pmatrix}
1 & 0  \\
0 & 0\end{pmatrix} +
 \frac{1}{3}\begin{pmatrix}
0 & 1  \\
1 & 1\end{pmatrix} (x_1 + x_2+x_3+x_4)
+ \begin{pmatrix}
0 & 0  \\
0 & 1
\end{pmatrix}  \sum_{ (u,v,t) \in \{a,b,c,d\}  } x(u,v,t)  \succeq 0
  \end{equation*}
  \begin{equation*}
3+Q_{n,k}(a)x_1+Q_{n,k}(b)x_2 + Q_{n,k}(c)x_3 +Q_{n,k}(d)x_4 \geq 0, \quad k=1,2,\dots,p
  \end{equation*}
   \begin{align*}
&S^n_k(1,1,1)+S^n_k(a,a,1)x_1+S^n_k(b,b,1)x_2+S^n_k(c,c,1)x_3
  +S^n_k(d,d,1)x_4 \\
&+ \sum_{ (u,v,t) \in \{a,b,c,d\}  } S^n_k(u,v,t)x(u,v,t) \succeq 0, \quad k=0,1,
\dots, p  \nonumber
\end{align*}
$$x_1,x_2,x_3,x_4,x(u,v,t)\geq 0,$$
where $S_k(\cdot,\cdot,\cdot)$ are $(p-k+1)\times(p-k+1)$ matrices with entries related to Gegenbauer polynomial and explicit definition can be found in \cite{BY2013} equation (2-3).
\end{theorem}
In this theorem the variables $x_1,\ldots, x_4$ refer to the number of ordered pairs of vectors in $Y$ with inner product $a,b ,c $ and $d$ respectively; for instance
  $(1/3)x_1=|Y|^{-1}\big|\{(c_1,c_2)\in Y^2: \langle c_1,c_2\rangle=a\}\big|.$
The variables $x(u,v,t)$ refer to the number of triples in $Y$ such that inner products are $(u,v,t)$, where $(u,v,t) \in \{ a, b, c, d\}$.
The number of  $(u,v,t)$ is the combinations with repetition for choosing three times out of 4 elements  and it is ${4+3-1 \choose 3} =20$.
We use CVX solver in MATLAB to solve the above optimization problems.
 We set up $p=8$ i.e.\ $S^n_0$ matrix is an $9\times 9$ matrix.
 The key inequalities to set up SDP bounds for spherical few distance sets are:
$$\sum_{c_i, c_j \in Y} Q_{n,k} ( \langle c_i, c_j \rangle) \geq 0 $$ and $$\sum_{c_i, c_j,c_m  \in Y} S^n_k ( \langle c_i, c_j \rangle, \langle c_i, c_m \rangle, \langle c_j, c_m \rangle) \succeq 0. $$

 Therefore, for spherical $s$-distance sets with $s\geq 5$, it is not hard to set up the SDP upper bounds for the cardinality and clarify the feasibility of our SDP bounds.



\section{Tight spherical designs of harmonic index $\{6,4\}$, $\{6,2\}$, $\{8,6\}$, $\{8,2\}$, as well as $\{10,6,2\}$, $\{12,8,4\}$}\label{sec:several}
$ $\\
\noindent
We consider the cases when $T=\{t_1,t_2,\ldots,t_{\ell}\}$ and $L(x)$ takes the minimum value $-c_{n,T,L}$ at $\ell$ non-negative points, where \[ L(x) = Q_{n,2e}(x) + f_{t_2}Q_{n,t_2}(x) + \cdots + f_{t_{\ell}}Q_{n,t_{\ell}}(x). \]
In this section, we will prove the non-existence of tight spherical designs of harmonic index $T$ for some $T$ with $\ell=2$ or $3$.

\subsection{The non-existence of tight spherical designs of harmonic index $\{6,4\}$}
$ $\\
Find $f_4$ such that
 \begin{equation}\label{equ:7.3}
Q_{n,6}(x)+f_4Q_{n,4}(x)=ax^2(x^2-\alpha^2)^2-c_{n,T,L}.
\end{equation}
By comparing the coefficients in \eqref{equ:7.3}, we get the following results:
\[
\begin{array}{l}
 \ds \alpha^2=\frac{15}{2(n+8)}-\frac{15}{(n+8)(n+10)}f_4,\\
\\
\ds \alpha^4=\frac{45}{(n+6)(n+8)}-\frac{180}{(n+4)(n+8)(n+10)}f_4.
\end{array}
\]
Solving for $f_4$ and $\alpha^2$ from these two equations gives:
\[ \begin{array}{l}
f_4= \dfrac{n+10}{10(n+4)(n+6)} \left( (n+6)(n-12) \pm 2(n+8)\sqrt{-(n+6)(n-4)} \right) , \\
\\
\alpha^2 = \dfrac{3\left( 2(n+6) \pm \sqrt{-(n+6)(n-4)} \right)}{(n+4)(n+6)}.
\end{array} \]
If $n \geq 5$, then $f_4$ is a complex number.
So we cannot find $L(x)=Q_{n,6}(x)+f_4Q_{n,4}(x)$ satisfying our assumption.
It is easy to check that $b_{2,T}=2$, $b_{3,T}=3$ (or 0) and $b_{4,T}=2$.
By Remark~\ref{remark:4.2}, there exists no tight spherical design of harmonic index $\{6,4\}$ when $n=2$ and $\left| Y \right|=2$.
When $n=3$ and $n=4$, observe that the lower bounds for spherical designs of harmonic index $6$  are about $3.41$ and $5.29$, respectively, which are strictly larger than $b_{3,T}$ and $b_{4,T}$.
Note that spherical design of harmonic index $\{6,4\}$ should also satisfy the condition for harmonic index $6$.
From the discussion above, there exists no tight spherical design of harmonic index $\{6,4\}$ for any $n$.

\subsection{The non-existence of tight spherical designs of harmonic index $\{6,2\}$}
$ $\\
The Gegenbauer polynomial $Q_{n,2}(x)$ is
\[ Q_{n,2}(x)=\frac{(n+2)(nx^2-1)}{2}. \]
Let $Q_{n,6}(x)+f_2Q_{n,2}(x)=ax^2(x^2-\alpha^2)^2-c_{n,T,L}$.
Then with similar calculation we have the following results.
\[ \begin{array}{l}
{\displaystyle f_2=\frac{(n-2)(n+4)(n+10)}{32(n+8)}, \quad \alpha=\pm \sqrt{\frac{15}{2(n+8)}}}, \\ \medskip
%{\displaystyle c_{n,T}=\frac{(n+2)(n+6)(n+10)(7n-4)}{192(n+8)}},\\ \\
{\displaystyle b_{n,T}=\frac{n(n+4)(2n+1)^2}{15(7n-4)}}\\ \medskip
{\displaystyle \quad \quad = \frac{1}{2401 \times 15}(1372n^3+7644n^2+10199n+7200)+ \frac{1920}{2401(7n-4)}}.
\end{array} \]
\noindent \textbf{Case (1)}:
 If $n \geq 8817$, i.e., $\left| \frac{1920}{2401(7n-4)}\right|< \frac{1}{2401 \times 15}$, then  $b_{n,T}$ is not an integer.

\noindent \textbf{Case (2)}: Tight spherical design $Y$ of harmonic index $\{6,2\}$ is regarded as an at most 3-distance set in $\Sphere^{n-1}$ with $I(Y) \subseteq \{0,\pm \alpha\}$.\\
Lemmens-Seidel (1973) proved the following fact.
\begin{theorem}[{\cite[Theorem 3.4]{LS1973}}]
If there are $|X|$ equiangular lines with angle $\arccos$ $\alpha$ in Euclidean $n$-dimensional space $\mathbb{R}^n$, and if $|X| > 2n$, then $1/ \alpha$ is an odd integer.
\end{theorem}
Then we can give a weaker condition for $\alpha$ with $3$-distance set as follows.
\begin{theorem}\label{theo:7.2}
If $X \subseteq \Sphere^{n-1}$ is a spherical $3$-distance set with $I(X) = \{0, \pm \alpha\}$ and $\left| X \right| > 2n$, then $1/ \alpha$ is an integer.
\end{theorem}

\begin{proof}
Let $X$ be a set of unit vectors whose mutual inner product set is $\{0, \pm \alpha\}$, and let $G$ be the Gram matrix of such vectors.
Then,
\begin{equation*}
G=\left(\begin{array}{cccc}
1 & & x \\
& \ddots &\\
x & & 1
\end{array}\right), \quad
A=\frac{1}{\alpha}(G-I)=
\left(\begin{array}{cccc}
0 & & x^{\prime} \\
& \ddots &\\
x^{\prime} & & 0
\end{array}\right),
\end{equation*}
where $x \in \{0, \pm \alpha\}$ and $x^{\prime} \in \{0, \pm 1 \}$.

$G$ is a symmetric and positive semidefinite matrix of order $|X|$.
It has the smallest eigenvalue $0$ of multiplicity $m \geq |X|-n$.
Therefore, $A$ has the smallest eigenvalue $-1/ \alpha$ of multiplicity $m \geq |X|-n$.
Moreover, $-1/ \alpha$ is an algebraic integer since $A$ is an integer matrix, and every algebraic conjugate of $-1/ \alpha$ is also an eigenvalue of $A$ with multiplicity $m$.
If $|X| > 2n$, then $m > \frac{|X|}{2}$.
Note $A$ cannot have more than one eigenvalue of multiplicity $m$ because $A$ is a $|X| \times |X|$ matrix.
Therefore $-1/ \alpha$ is rational, since it is also an algebraic integer, hence $-1/ \alpha$ is an integer.
\end{proof}
There is an example of $X$ in $S^{n-1}$ with $I(X) = \{0, \pm \alpha\}$ and $|X| > 2n$ so that $1/\alpha$ is an even integer.
\begin{example}
The $E_8$ root system consists of 240 points in $S^7$ with inner products $0,\pm 1/2,-1$. An example of the above is a half of $E_8$-roots, which consists of 120 points in $S^7$ by choosing one of antipodal pairs.
\end{example}
Assume $X$ is a spherical 3-distance set with $I(X)=\{0,\pm\alpha\}$ such that $|X|\geq |Y|$.
If $5 \leq n \leq 8816$, then $|X|\geq |Y| >2n$.
By Theorem~\ref{theo:7.2}, we know that $1/ \alpha \in \Integer$.
And it is easy to check that $\frac{1}{\alpha}=\sqrt{\frac{2(n+8)}{15}} \in \Integer$ and $b_{n,T} \in \Integer$ cannot hold simultaneously for $5 \leq n \leq 8816$.

\noindent \textbf{Case (3)}:
 If $2 \leq n \leq 4$, then $b_{2,T}=2$ is the unique integral case. By Remark~\ref{remark:4.2}, $Y=\{(1,0), (0,1)\}$ is a tight spherical design of harmonic index $\{6,2\}$ in $\Sphere^1 \subseteq \Real^2$.

\subsection{The non-existence of tight spherical designs of harmonic index $\{8,6\}$}
$ $\\
 Find $f_6$ such that
\begin{equation}\label{equ:7.2}
Q_{n,8}(x)+f_6Q_{n,6}(x)=a(x^2-\alpha^2)^2(x^2-\beta^2)^2-c_{n,T,L}.
\end{equation}
Set $\gamma=\alpha^2+\beta^2$, $\xi=\alpha^2 \beta^2$.
By comparing the coefficients in \eqref{equ:7.2}, we get the following results:

 \begin{equation}\label{equ:7.3}
\ds \gamma=\frac{14}{n+12}-\frac{28}{(n+12)(n+14)}f_6, 
 \end{equation}
\begin{equation}\label{equ:7.4}
 \gamma^2+2\xi=\frac{210}{(n+10)(n+12)}-\frac{840}{(n+8)(n+12)(n+14)}f_6,
\end{equation}
\begin{equation}\label{equ:7.5}
\small \gamma\xi=\frac{210}{(n+8)(n+10)(n+12)}-\frac{1260}{(n+6)(n+8)(n+12)(n+14)}f_6.
\end{equation}
Using Eq.~\eqref{equ:7.3} and Eq.~\eqref{equ:7.5}, we have the following relations:
\[
{\ds f_6=\frac{n+14}{2}\left(1-\frac{n+12}{14}\gamma \right)}, \quad
{\ds \gamma=\frac{420}{(n+10)\big(45-(n+6)(n+8)\xi\big)} }.
\]
Therefore Eq. \eqref{equ:7.4} and the above two relations imply that 
\begin{eqnarray}\label{equ:cubic}
&&  (n+6)^2(n+8)^3(n+10)^2\xi^3\nonumber\\
&&+15(n+6)(n+8)^2(n+10)(n-18)\xi^2 \nonumber\\
&&-225(n+8)(n+10)(5n-6)\xi+1575(11n-2)=0.
\end{eqnarray}
We denote the LHS of \eqref{equ:cubic} as $g(\xi)$, then $g(\xi)$ must have at least one non-negative root since we define $\xi= \alpha^2\beta^2$. 
However we can prove that $g(\xi)$ only has one real root and that root is negative. $g(\xi)$ is a degree 3 polynomial with positive leading coefficient and has two critical points:
 $$\xi_1=-\frac{5(5n-6)}{(n+6)(n+8)(n+10)}, \qquad \xi_2=\frac{15}{(n+6)(n+8)}.$$
Therefore, for $n\geq 2$, $g(\xi)$ has local maximum at $\xi_1<0$ and local minimum at $\xi_2>0$.
We have $g(\xi_2)=\frac{7200(n+3)(n-4)}{n+6}>0$ if $n>4$.
Then, $g(\xi)$ only has one real root which is negative for $n\geq 5$. 

From the calculation, we obtain the lower bound for $\left| Y \right|$ is
\[ b_{n,T}=\frac{1}{252 \times 9.427094401\cdots} \times n^4(1+o(1)).\]
If $2 \leq n \leq 4$, it is easy to check that $b_{4,T}=2$ is the unique case when $b_{n,T} \in \Integer$. However, the lower bound for spherical design of harmonic index $6$ is about $5.29 $, which is strictly larger than $b_{4,T}$.
Therefore, there is no tight spherical design of harmonic index $\{8,6\}$.
\subsection{The non-existence of tight spherical designs of harmonic index $\{8,2\}$}
$ $\\
We want to determine $f_2$ such that
\[Q_{n,8}(x)+f_2Q_{n,2}(x)=a(x^2-\alpha^2)^2(x^2-\beta^2)^2-c_{n,T,L}.\]
From calculation we have the following results:
\[f_2=\frac{(n-2)(n+4)(n+5)(n+6)(n+14)}{90(n+12)^2},\]
\[\alpha^2,\beta^2=\frac{7}{n+12} \pm \frac{\sqrt{42(n+5)(n+10)}}{(n+10)(n+12)}.\]
%\[c_{n,T}=\frac{(n+2)(n+4)(n+5)(n+8)(n+14)(n^3+27n^2+356n-240)}{240(n+10)(n+12)^3}.\]
Then we have
\[ \begin{array}{rcl}
b_{n,T} & = & {\displaystyle \frac{n(n+6)(n+5)(n^2+15n+8)^2}{168(n^3+27n^2+356n-240)}} \\
& & \\
&=& {\displaystyle \frac{1}{168}(n^4+14n^3-133n^2+2638n-10584)} \\
& & \\
& & {\displaystyle +\frac{-4032n^2+26208n-15120}{n^3+27n^2+356n-240}}.
\end{array} \]
Let $p(n)=-4032n^2+26208n-15120$ and $q(n)=n^3+27n^2+356n-240$.
Then $|\frac{p(n)}{q(n)}| <\frac{1}{168}$ gives a condition for $b_{n,T}$ not being an integer.
%$\frac{p(n)}{q(n)} <\frac{1}{168}$ if $n \leq 6$ and $\frac{p(n)}{q(n)} > -\frac{1}{168}$ if $1 \leq n \leq 5$ or $n \geq 677343$.
This implies that $b_{n,T}$ cannot be an integer if $n \geq 677343$.
We can check the remaining cases where $b_{n,T}$ is an integer, and obtain
$b_{2,T}=2$, $b_{4,T}=9$, $b_{9,T}=96$.
The case $b_{2,T}=2$ is eliminated by Remark~\ref{remark:4.2}.

If $n=4$, the SDP upper bound for $4$-distance set is $8.9981$ ($<9$).

For $n=9$, we assume $X$ is a spherical 4-distance set with $I(X)=\{\pm\alpha,\pm\beta\}$ such that $|X|\geq |Y|$.
Then $|X|\geq |Y|=b_{n,T} \geq  2\binom{n+1}{2}$.
But neither $\frac{1-\alpha^2}{\beta^2-\alpha^2}$ nor $\frac{1-\beta^2}{\alpha^2-\beta^2}$ is an integer.
By Lemma~\ref{lem:rat}, the case where $n=9$ is also impossible.

\subsection{The non-existence of tight spherical designs of harmonic index $\{10,6,2\}$.} \label{sec:10-6-2}
$ $\\
Consider the case where $L(x)=Q_{n,10}(x)+f_6Q_{n,6}(x)+f_2Q_{n,2}(x)$ takes minimum value $-c_{n,T,L}$ at three non-negative points $\{0,\alpha,\beta\}$, i.e.,
\[ L(x) = Q_{n,10}(x)+f_6Q_{n,6}(x)+f_2Q_{n,2}(x)  =  ax^2(x^2-\alpha^2)^2(x^2-\beta^2)^2-c_{n,T,L}.\]
We can solve for $f_6$ and $f_2$ as follows.
\[ \begin{array}{l}
{\displaystyle f_6=\frac{(n-2)(n+8)(n+18)(13n+28)}{1344(n-8)(n+16)}}, \\
\\
{\textstyle f_2=\frac{(n-2)(n+4)(n+8)(n+14)(n+18)(37n^3-742n^2+1792n+20256)}{129024(n-8)^2(n+12)(n+16)}}.
\end{array} \]
Then we can get $\alpha^2,\beta^2$ and the lower bound $b_{n,T}$.
\[ \begin{array}{l}
{\displaystyle \alpha^2,\beta^2=\frac{45(n-8)(n+12) \pm \sqrt{15(n-8)(n+12)(43n^2-244n-1952)}}{4(n-8)(n+12)(n+16)}},\\
\\
{\displaystyle b_{n,T}=\frac{ n (n+4)(n+8) (4 n^3-10 n^2-143 n-84)^2}{45(781n^4-9548n^3+10128n^2+160960n-108032)}}.
\end{array} \]
The following theorem is very useful to consider the existence of spherical design of harmonic index $\{10,6,2\}$.
\begin{theorem}[{\cite[Theorem 5.3]{N2011}}]\label{theo:7.3}
Let $X$ be an antipodal $s$-distance set on $\Sphere^{n-1}$ where $s$ is an even integer at least $4$. Let $I(X)=\{-1, \beta_1=0, \pm\beta_2, \ldots, \pm\beta_{\frac{s}{2}} \}$.
\begin{enumerate}[(1)]
\item
 Let $N=\binom{n+s-3}{s-2}$. If $\left| X \right| \geq 4N$, then for each $i = 1,\dots,s/2$,
\[ k_i := \prod_{j=1,\dots,\frac{s}{2}, j  \neq i} \frac{1 - \beta_j^2}{\beta_i^2 - \beta_j^2} \]
must be an integer with $\left| k_i \right| \leq U(N)$.\\
\item
 Let $N=\binom{n+s-4}{s-3}$. If $\left| X \right|\geq 4N+2$, then for each $i = 1,\dots,s/2$,
\[ k_i := \frac{1}{\beta_i}\prod_{j=2,\dots,\frac{s}{2}, j  \neq i} \frac{1 - \beta_j^2}{\beta_i^2 - \beta_j^2} \]
must be an integer with $\left| k_i \right| \leq \lfloor \sqrt{2N^2/(N+1)} \rfloor $.
\end{enumerate}
\end{theorem}

Similarly, tight spherical design of harmonic index $\{10,6,2\}$ is regarded as an at most 5-distance set $Y \in \Sphere^{n-1}$  with $I(Y)\subset\{0, \pm\alpha, \pm\beta\}$.
We construct an antipodal set $X^{\prime}=Y \cup (-Y)$ with $I(X^{\prime})\subset\{-1,0, \pm\alpha, \pm\beta\}$.

In this subsection, we assume $X$ is an antipodal spherical 6-distance set with $I(X)=\{-1,0, \pm\alpha, \pm\beta\}$ such that $|X|\geq |X^\prime|=2|Y|$.\\
\noindent \textbf{Case (1)}:
 If $n \geq 170$, then $|X|/2\geq |Y| \geq 2\binom{n+6-3}{6-2}+1$.
Theorem~\ref{theo:6.3} implies that $\alpha,\beta \in \Rational$. Namely, $\alpha\beta=\sqrt{\frac{(n-8)(n+12)(n+16)}{15(23n-172)}} \in \Rational$.
So there exists an integer $u$ such that
$$15(n-8)(n+12)(n+16)(23n-172)=u^2.$$
Assume that $n-8 = Ay_1^2$, $n+12 = By_2^2$, $n+16 = Cy_3^2$, $23n-172 = Dy_4^2$.
Let $p(n_1,n_2)$ be a prime divisor of $\gcd(n_1,n_2)$.
Since $\gcd(n-8,n+12)$ can be one of $1,2,4,5,10,20$, then $p(n-8,n+12)=2$ or $5$. And with similar argument, we obtain the following result.
\[ \begin{array}{ll}
p(n-8,n+12)=2\ \text{or} \ 5, & p(n-8,n+16)=2 \ \text{or} \ 3, \\
\\
  p(n-8,23n-172)=2 \ \text{or} \ 3, & p(n+12,n+16)=2,  \\
\\
   p(n+12,23n-172)=2 \ \text{or} \ 7, & p(n+16,23n-172)=2, \ 3, \text{or} \ 5.
\end{array} \]
Then we get $16$ elliptic equations
\[ (n-8)(n+12)(n+16)=E y^2 \]
with $E \mid 2 \times 3 \times 5 \times 7$. We can obtain the integral solutions of these equations.

Multiply both sides of the equation $(n-8)(n+12)(n+16)=E y^2$ by $E^3$ and make linear transformation $M=E^2y$ and $N=E(n+12)$. Then $(n-8)(n+12)(n+16)=E y^2$ becomes $N(N-20E)(N+4E)=M^2$ with coefficients of $M^2$ and $N^3$ being $1$.

If $E=2$, with the linear transformation $N=2n+24$ and $M=4y$, the equation becomes $N(N-40)(N+8)=M^2$. It has only three integral solutions $(N,M)=(0,0)$, $(40,0)$ and $(-8,0)$.

In the table below, we list all the cases when the equation $(n-8)(n+12)(n+16)=E y^2$ has non-trivial integral solutions.

\[ \begin{array}{c|l}
E & \text{solution}(n,y) \\ \hline%&
5 &  (33,105)  \\  \hline
7  & (16,32),(68,240) \\ \hline
6  & (20,48), (48,160)  \\ \hline
10  &  (24,48),(38,90)  \\ \hline
15  &  (308,1440)  \\ \hline
21   & (9,5),(488,2400)  \\ \hline
42   &  (12,8),(128,240)  \\ \hline
 210   &  (44,24) \\ \hline
\end{array} \]
There are two values for $n \geq 170$, i.e., $n=308,488$. But in these two cases, $\alpha \notin \Rational$.

\noindent \textbf{Case (2)}:
If $19 \leq n \leq 169$, then the second statement in Theorem \ref{theo:7.3} implies that $\frac{1-\beta^2}{\alpha(\alpha^2-\beta^2)}$ must be an integer. And it is easy to check this cannot be satisfied.

\noindent \textbf{Case (3)}:
 If $2 \leq n \leq 18$, then $b_{8,T}=8$ is the only integral case. Let
\[ m_1=\alpha^2+\beta^2, ~ m_2=(\alpha^2+\beta^2)^2+2\alpha^2\beta^2, ~ m_3=\alpha^2\beta^2(\alpha^2+\beta^2). \]
Then $m_1(m_2-m_1^2)-2m_3=0$. However, comparing the coefficients in $L(x)$ and $R(x)$, we obtain
\[
{\textstyle
\frac{1}{2}m_1(m_2-m_1^2)-m_3=\frac{1}{(n+12)(n+14)(n+16)^2} \left( \frac{18900(n-8)}{(n+8)(n+18)}f_6-\frac{225(13n+28)(n-2)}{16(n+16)} \right).}
\]
If $n=8$, then the RHS of the above equation is $-\frac{15}{8192} \neq 0$.
This is a contradiction. And it implies that, when $n=8$, we cannot find a function $L(x)$ satisfying our assumption.

\subsection{Tight spherical designs of harmonic index $\{12,8,4\}$.}
$ $\\
Consider the case where $L(x)=Q_{n,12}(x)+f_8Q_{n,8}(x)+f_4Q_{n,4}(x)$ takes minimum value at three positive points $\{\alpha,\beta,\gamma\}$, i.e.,
\[
\begin{array}{l}
F(x)=Q_{n,12}(x)+f_8Q_{n,8}(x)+f_4Q_{n,4}(x)+c_{n,T,L}\\
\\
\qquad =a(x^2-\alpha^2)^2(x^2-\beta^2)^2(x^2-\gamma^2)^2=R(x),
\end{array}
\]
where $c_{n,T,L}=-\min\limits_{-1\leq x\leq 1}L(x)$.

Let $Z_1=\alpha^2+\beta^2+\gamma^2$, $Z_2=\alpha^2\beta^2+\alpha^2\gamma^2+\beta^2\gamma^2$ and $Z_3=\alpha^2\beta^2\gamma^2$.
Comparing the coefficients in $F(x)$ and $R(x)$, we can obtain the following results.
$$m_1=Z_1, \ m_2=Z_1^2+2Z_2,\ m_3=Z_1Z_2+Z_3,\ m_4=Z_2^2+2Z_1Z_3,\ m_5=Z_2Z_3,$$
where $-2am_1$, $am_2$, $-2am_3$, $am_4$ and $-2am_5$ are the coefficients of $x^{10}$, $x^8$, $x^6$, $x^4$, $x^2$ in polynomial $R(x)$, respectively.
 Note that $m_i$ is a linear combination of $f_8$ and $f_4$.
 Substituting $Z_2=\frac{m_2-m_1^2}{2}$ and $Z_3=m_3-Z_1Z_2=m_3-m_1(\frac{m_2-m_1^2}{2})$ into $m_4$ and $m_5$, we can solve for $f_8$ and $f_4$ as follows.
\[ \begin{array}{l}
{\displaystyle f_8=\frac{(n+8)(n+9)(n+22)}{180(n+20)}},\\
\\
{\displaystyle f_4=\frac{(n+4)(n+5)(n+8)(n+9)(n+18)(n+22)}{7200(n+16)(n+20)}}.
\end{array} \]
Immediately we have
\[ \begin{array}{l}
{\displaystyle Z_1=\alpha^2+\beta^2+\gamma^2=\frac{33}{n+20}},\\
\\
{\displaystyle Z_2=\alpha^2\beta^2+\beta^2\gamma^2+\alpha^2\gamma^2=\frac{231}{(n+16)(n+20)}},\\
\\
{\displaystyle Z_3=\alpha^2\beta^2\gamma^2=\frac{231}{(n+12)(n+16)(n+20)}},\\
\\
{\displaystyle b_{n,T,F}=\frac{1}{27720}(n+1)(n+2)(n+5)(n+6)(n+9)(n+10)}.
\end{array} \]
We should remark that there is another test function $F_1(x)=Q_{n,12}(x)+f_8^{\prime}Q_{n,8}(x)+f_4^{\prime}Q_{n,4}(x)+f_0^{(1)}$, where
\[{\scriptstyle f^{\prime}_8= -\frac{(n+16)(n+22)(n^3-19n^2-1564n-4064)}{220(n-4)(n+8)(n+20)}},\]
\[{\scriptstyle f^{\prime}_4= \frac{(n+4)(n+18)(n+22)(73n^6-1238n^5-31603n^4+304204n^3+5781440n^2+22288384n+25661440)}{61600(n-4)^2(n+8)(n+20)^2}}.\]
 And in this case we have the following lower bound.
\[ \begin{array}{lll}
b_{n,T,F_1}&=&{\scriptstyle \frac{1}{216}(n+2)(n+6)(n+10)(n^6+39n^5+407n^4+989n^3+56556n^2+279424n+357888)^2 \times }\\ \smallskip
&& {\scriptstyle (955n^9-48483n^8+1094513n^7+5943943n^6-158073240n^5+1292462960n^4+}\\ \smallskip
&& {\scriptstyle 29035332352n^3+132584060928n^2+227750248448n+126509383680)^{-1}}.
  \end{array}\]
It is easy to check that $b_{n,T,F_1}>b_{n,T,F}$ if and only if $30 \leq n \leq 38$.
Then
$$b_{n,T}=\left\{
\begin{array}{ll}
b_{n,T,F_1} & \text{if~}  30 \leq n \leq 38,\\ [0.1in]
b_{n,T,F} & \text{otherwise.}
\end{array}
\right.$$
Assume $X \in \Sphere^{n-1}$ is an antipodal spherical 7-distance set with
$I(X)=\{-1,\pm\alpha,\pm\beta,\pm\gamma\}$ such that $|X|\geq 2|Y|$, where $Y$ is a tight spherical design of harmonic index $\{12,8,4\}$.\\
\noindent \textbf{Case (1)}:
If $n \geq 439$, then $|X|/2\geq |Y|=b_{n,T} \geq 2\binom{n+7-3}{7-2}+1$. Theorem \ref{theo:6.3} implies that
\[ \alpha \beta \gamma=\sqrt{\frac{231}{(n+12)(n+16)(n+20)}} \in \Rational .\]
Equivalently, we have $(n+12)(n+16)(n+20)=231y^2$ for some integer $y$.
With linear transformation $M=231^2y$ and $N=231(n+16)$, we have $N(N^2-462^2)=M^2$.  It has integral solutions
$(N,M)=(-528,17424)$, $(-252,14112)$, $(1617,53361)$, $(3388,189728)$. However, this gives no positive integral solution for $n=N/231-16$.

 \noindent \textbf{Case (2)}:
 If $34 \leq n \leq 438$, then $|X|/2\geq |Y|=b_{n,T} \geq 2\binom{n+7-4}{7-3}+1$. By Theorem~\ref{theo:7.3}, $\frac{(1-\beta^2)(1-\gamma^2)}{\alpha(\alpha^2-\beta^2)(\alpha^2-\gamma^2)}$ must be an integer. And it is easy to check this cannot be satisfied.

 \noindent \textbf{Case (3)}:
 If $n \leq 33$, then $b_{5,T}=35$, $b_{6,T}=64$, $b_{9,T}=285$, $b_{13,T}=1311$, $b_{16,T}=3315$, $b_{20,T}=9425$, $b_{23,T}=18560$ are the integral cases.
  But LP upper bounds for spherical $6$-distance set corresponding to $n=5,6,20,23$ are $30.2656$, $59.8173$, $9405.11$, $17926.1$, respectively.
  When $n=9,13,16$, we set up the SDP method on the upper bounds for spherical $6$-distance sets.
  However, the upper bounds coincide with our lower bounds $b_{n,T}$.
  We conclude that there exists no tight spherical design in $\Sphere^{n-1}$ of harmonic index $\{12,8,4\}$ if $n \neq 9,13,16$.
The existence of such tight designs for $n=9,13,16$ is still open.

\section{Concluding remarks}
\label{sec:remark}
\noindent
In this paper we considered mainly spherical designs of harmonic index $T=\{t\},$ or $T=\{t_1,t_2\}$.
For some $T=\{t_1,t_2,\ldots ,t_{\ell}\}$ (with $t_1=2e>t_2> \cdots > t_{\ell}$ and all $t_i$ are even), it seems that the general interesting case is where $L(x)=Q_{n,2e}(x)+f_{t_2}Q_{n,t_2}(x)+f_{t_3}Q_{n,t_3}(x)+\cdots +f_{t_{\ell}}Q_{n,t_{\ell}}(x)$ and the minimum value of $L(x)$ is at $\ell$ non-negative points $\alpha_1, \alpha_2,  \ldots , \alpha_{\ell}$.
Thus, further studies along this line would be interesting.

As we have shown in Section~\ref{sec:asymptotic} as well as in previous sections, it seems remarkable that a spherical design of harmonic index $t=2e$ has a Fisher type lower bound $\left| Y \right|\geq$ (constant)$\cdot n^e$, which is the same order as for spherical $2e$-design.
So, all harmonic index $T$-designs are between harmonic index $2e$-designs and spherical $2e$-designs.
It seems that considering tight $T$-designs have some meaning, although it seems tight harmonic index $T$-designs rarely exist.

As it is discussed in Bannai-Okuda-Tagami \cite[Proposition 4.1]{BOT2015} that to some extent,
\[b_n := \lim_{e\rightarrow \infty} b_{n,2e}\]
is also studied.
The result is explained in terms of Bessel functions.
As some special cases are mentioned in \cite[p.~10]{BOT2015}, $b_n$ becomes greater than $n(n+1)/2$ if $n=7,8,9,10$.
This implies that tight spherical designs of harmonic index $2e$ do not exist, if $t=2e$ becomes large, say for $n=7,8,9,10$.
On the other hand if $n\leq 6$, it seems possible to determine the non-existence of tight designs in these cases, but it is not clear how we can show the non-existence of such harmonic index $2e$-designs whose sizes are close to the Fisher type lower bound.
It seems that this remains as an interesting open problem.

In concluding this paper, we remark that the theory as well as the concept of harmonic index $T$-designs in Q-polynomial association schemes exactly go parallel with the spherical case.
The concept of $T$-design for an arbitrary subset $T$ of the index set of nontrivial relations $\{1,2, \ldots ,d\} $ is already defined in Delsarte \cite[Section 3.4]{D1973} (1973).
On the other hand, it seems that any systematic study on some specific choices of $T$, beyond the case $T=\{1,2, \ldots t\}$ has not begun, even for the case $T=\{t\}$.
We hope to discuss more on this topic in a separate paper.

\section*{Acknowledgments}
We thank Takayuki Okuda and Makoto Tagami for many helpful discussions on the earlier stage of the study of spherical designs of harmonic index $T$, in particular when $T=\{8,4\}.$ We thank Ziqing Xiang, Shunichi Yokoyama, Masanobu Kaneko, William Stein and Maosheng Xiong for the discussions of the elliptic diophantine equations, and the help how to use the database. Eiichi Bannai is supported in part by NSFC grant No. 11271257.








% \bibliographystyle{plain} 
% \bibliography{myBibFile} 
% If you use BibTeX to create a bibliography
% then copy and past the contents of your .bbl file into your .tex file



\end{document}
%%% Local Variables:
%%% mode: latex
%%% TeX-master: t
%%% End:
